# **Extract of Chapter Conclusions**

Thesis Title: A critical analysis of the Australian government's rationale for its vaccination policy
Judy Wilyman
University of Wollongong
2015

Link to full document: https://ro.uow.edu.au/cgi/viewcontent.cgi?article=5550&context=theses

**Chapter 2 Conclusion** 

**Chapter Title: Controlling Infectious Diseases in Australia** 

Chapter pages: 11-44

## 2.9 Conclusion

The epidemiological triad illustrates that infectious diseases are a result of the interaction of multiple causal factors in pathogenesis. In other words, infectious agents (bacteria/viruses) are necessary but do not always result in disease; exposure to these agents can result in a diversity of health outcomes. This is why the assessment of the severity of each infectious agent in the community cannot be made outside the ecological context. Global public health policies are resulting in a 'one size fits all' vaccination program that ignores the context in which infectious diseases occur and the diversity of outcomes that are expected. There are a range of health outcomes that can arise after exposure to an agent – no disease, mild or severe disease, or death - and the health outcome is dependent upon the host, environment and agent characteristics. Humans develop immunity by adapting to their environment through a process known as homeostasis. Homeostasis is part of the process by which humans develop resistance

(immunity) to disease and it develops through natural interactions and exposure with the organisms in an ecological context. The immune system interacts with the environment and other body systems in complex ways to develop long-term protection against disease: it is not simply the production of antibodies to an antigen in the process of seroconversion induced by vaccination. In this way communities are protected by herd immunity developed by natural exposure to the agents.

The infant mortality rate in Australia declined primarily due to improvements in the ecological context including sanitation, hygiene, nutrition, breast feeding, smaller family sizes, less crowding and improved infrastructure. The risk of death and illness due to infectious diseases had declined by 1950 before most mass vaccination programs were introduced. In 1950 only two vaccines were in voluntary use in mass vaccination programs in Australia, diphtheria and smallpox, and these diseases did not decline any more quickly than other infectious diseases. Whilst outbreaks of infectious diseases still occurred after 1950, the risk of death and severe illness was low to negligible to the majority of Australian children and long-term herd immunity to the diseases was gained by natural infection during childhood. Many mass vaccination campaigns were introduced in Australia from 1952-1990 but participation was voluntary and without coercive government strategies. Infectious diseases continued to decline as living standards and education improved. Despite the low infant mortality rate in the early 1990's and the lack of significance of infectious diseases in Australia at this time, the government implemented a new strategy to increase the vaccination rates in the population and to expand the recommended schedule of vaccines. This policy change was in response to WHO global health directives, and not a specific recommendation for the ecological context of the Australian situation.

A decision was made in the mid-twentieth century to medicate all children to prevent disease, despite the diversity of outcomes that arise after exposure to an infectious agent. At this time the health of Australians was measured using the decline in the infant mortality rate and since 1990 vaccination coverage has been used as the surrogate measure for 'health'. These surrogates have been used even though it is known that mortality rates and vaccination coverage are unable to inform authorities about the illness/disability associated with each disease. The well-being and quality of life of individuals in communities cannot be accurately measured using these surrogates. This

is relevant to vaccination policies because there has been a significant increase in chronic illness in Australian children that has occurred at the same time as the government increased the participation rates and the number of vaccines listed on the childhood vaccination schedule in the 1990's. Authorities that depend upon infant mortality rates and vaccination coverage alone to inform public health policy will not detect or recognize a correlation between vaccines and increased morbidity in the population.

#### **Overview of Chapters**

The expansion of the Australian government's vaccination program within the global framework for public health policy directed by the WHO/GAVI is described in chapter 3. Chapter 4 discusses the federal governance of the NIP and the way in which risk assessment for infectious diseases is performed and communicated to the public. A discussion of the principles of public health policies and ethical codes of conduct for health promotion is provided in chapter 5. Chapter 6 investigates the rigour of the science that is produced in academic-industry partnerships in research institutions and chapter 7 provides a discussion of the evidence the Australian government provides to support the claims used to promote vaccines to the public. Public health policies are a reflection of the cultural and political beliefs of the time. Chapter 8 discusses the political framework that results in *undone science* and how public health policy is being designed on political decisions made without complete scientific knowledge. It provides the political framework for the existence of undone science and explains the consequences of its existence in public policies to population health. These influences in the development of global vaccination programs are illustrated in a case study of the Human Papillomavirus (HPV) vaccine in chapter 9 and the 'Swine Flu' 2009 vaccine in chapter 10. An investigation of these vaccination programs demonstrates the political and economic environment in which vaccines are being produced in the globalization era and the effect this has on the research that is produced in academic institutions. Chapter 11 provides the conclusions drawn about the Australian government's claim that vaccines are a safe and effective prevention for many infectious diseases.

**Chapter 3 Conclusion** 

Chapter Title: Global Health Policy and Australia's National Immunisation Program (NIP)

Chapter pages:68-86

### 3.18 Conclusion

The Australian government's NIP has been set within the framework and directives of the WHO's global health policy. It has not been developed within the specific environmental context of the Australian community but to comply with directives from the WHO on global vaccination policies. These policies have been developed by WHO/UNICEF since the 1970's and inspired by the campaign to eradicate smallpox. Although there was disagreement at this time about the value of vaccination in eradicating diseases, WHO/UNICEF decided to expand the program to many other infectious diseases. This was initiated as the Expanded Program on Immunisation (EPI) and since the 1970's has developed through many phases in all WHO member countries, both developing and developed. The campaign has been promoted on the moral principle of 'saving the world's children with life-saving vaccines'. During the 1980's the influence of neoliberalism resulted in economic and political experts dominating the development of global health policies. Mixed health messages began to be presented through WHO directives as neoliberalism focused public health policies on technology-based interventions such as vaccination. This was done in many developing countries at the expense of primary healthcare programs that targeted the social and environmental determinants of health. In the 1990's the World Bank, the International Monetary Fund and the Rockefeller and Gates Foundations became partners with the WHO/UNICEF to sponsor global vaccination programs. A shortage of funding in the 1990's for the research and development of vaccines led to the development of publicprivate partnerships that were influential in the direction of global health policies in WHO member countries.

Economics and politics began to dominate the design of public health policies from this time, with health statistics obtained from non-governmental economic models of the cost-effectiveness of implementing vaccination programs. Global health directives were based on industry modeling for global communities instead of the specific ecological conditions of each country. In 2000 the direction of global public health policies changed with the establishment of the Global Alliance for Vaccines and Immunisation (GAVI), a body that consists of public-private partnerships with governments and corporations jointly influencing global health policy decisions. At this time vaccines became the sole focus of global policy: GAVI promoted vaccines to WHO member nations using financial incentives as well as by controlling the supply of vaccines to these countries. This resulted in the 'vaccine paradox' where governments, in alliance with corporations, control the supply and demand of vaccines. This represents a clear conflict of interest that is not transparent to the public. Since 2000 global public health policies have protected industry interests through the sole focus on vaccination programs and many authorities within the WHO, and developed nations, have expressed concern at these technology-based health programs. This has been at the expense of broader primary healthcare needs.

Australia's NIP expanded in the 1980's and 1990's according to WHO goals for achieving high participation rates for all the recommended vaccines. The goals were set to increase childhood vaccination rates to ninety percent for all the vaccines in Australia even though there was no significant threat to the majority of the population from the targeted infectious diseases. The decision to use vaccines for many diseases was a universal directive to both the developed and the developing countries, without risk/benefit assessments for the use of each vaccine in specific countries and populations. In addition, national governments did not provide an adequate surveillance system for the accurate determination of the frequency of causally related adverse events from vaccines. The harm caused by vaccines, either individually or in the combined schedule, has not been included in the economic modeling for the costeffectiveness of vaccination programs. These programs have been based on the assumption that vaccine-created herd immunity can prevent infectious diseases, if enough people participate in these programs, without causing significant harm to the population. The lack of empirical evidence for these assumptions is discussed in chapters 4 and 7.

During the 1990's government's re-labeled infectious diseases as vaccine-preventable diseases, a label that implies vaccines can prevent infectious diseases. The public has been informed through the mainstream media that high participation rates in vaccination campaigns are needed to *prevent* infectious diseases but the empirical evidence to support this claim has not been provided. The evidence the Australian government provides to support vaccination policies is discussed in Chapter 7. The government has introduced many new vaccines since the 1990's and emphasised the need for high vaccination rates for all infectious diseases for which there is a vaccine. New vaccines are continually added to the national recommended schedule. Each vaccine carries a risk to some individuals yet the Australian government does not provide a separate risk/benefit assessment for each disease and vaccine, nor for the combination of vaccines that are recommended in the children's schedule. The vaccines have been added to the recommended schedule without public consultation or participation in policy development, and without informing the public of the ingredients of vaccines. Financial incentives have been used to pressure parents and doctors to use all the recommended vaccines and to assist the government to track the vaccination status of children. These strategies were implemented in 1993 and formalized in 1997 as the Immunise Australia Program (IAP). These coercive practices have increased in 2015 with many employees now being required or expected to vaccinate even though the government continues to claim that vaccination in Australia is not compulsory.

The Australian Government's program does not make any reference to the historical evidence of the control of infectious diseases in Australia or the risk/benefit of using an increasing number of vaccines in children/adults. Mainstream media has been used since the 1990's to influence public behaviour by informing the public that high participation rates in vaccination programs are important in controlling infectious diseases without providing evidence for this claim. In contrast, the media was used in the early 20<sup>th</sup> century to successfully promote social (ecological) medicine to the public to reduce the threat of infectious diseases through environmental and lifestyle changes. The media message changed in the second half of the century when the focus of public health was directed to vaccination policies. Since this time the mainstream Australian media has emphasised the benefits of vaccines, without providing empirical evidence, and without informing the public of the known risks associated with each vaccine or the long-term health effects of the combined schedule of vaccines.

**Chapter 4 Conclusion** 

Chapter Title: Implementation of the Australian Government's

**Vaccination Policies** 

Chapter pages: 87-117

#### 4.11 Conclusion

Australia's National Immunisation Program (NIP) has been developed by the Federal government as a recommendation to the states and territories under the Public Health Acts. The public health laws were reformed in 1999 and the regulation of infectious diseases was placed under infectious disease law and separated from other public health issues - sanitation and environmental control – that were traditionally associated with infectious disease control. This resulted in the use of different methodologies for determining the health risk for infectious diseases and many other environmental health hazards. Risk assessment for infectious diseases is performed by computer modeling however for many other environmental health hazards, including the microbiological risk from food pathogens, it is determined using the Environmental Health Risk Assessment (EHRA) Guidelines. See Appendix 6. This framework for risk assessment is believed by environmental health practitioners to represent the most systematic and transparent method for assessing health risks for environmental hazards in genetically diverse communities.

Risk assessment for environmental hazards must include both technical and non-technical information. This is to address the limitations and uncertainty in the science. Gaps in scientific knowledge result in incorrect value judgments about safety and this can be countered with caution and non-technical input if the gaps in the science are acknowledged. The EHRA framework also ensures that the community is consulted regarding their perceptions of the risk of a hazard. A rigorous risk assessment can stand up to scrutiny by all stakeholders and the public must be encouraged to openly debate vaccination policies to maintain population health. The public perception of risk is essential to policy development to ensure that *health* is the primary focus of the policy. There are many variables and assumptions in computer modeling that are not made transparent to the public or government ministers. In addition, the precautionary principle (PP), with the onus of proof

on the proponent and not the general public, has not been adopted in infectious disease law. When the PP is not implemented with the onus of proof of harm on the proponent instead of the general public, it is possible for governments to claim there is no evidence of harm therefore no action needs to be taken, based on a lack of scientific evidence due to appropriate studies not being funded. The public cannot be provided with definitive evidence if the studies have not been funded.

A universal management strategy that carries a risk to sub-groups in the population needs to be openly debated by the community. The dominance of scientific experts in vaccination policy is questionable because of the increasing gaps in scientific knowledge due to undone science. This can synchronise with a lack of transparency in risk assessment to compromise health management strategies. A dependency on scientific experts and elite knowledge from computer models results in the exclusion of public participation in policy debate. Yet the existence of unfunded research in the underpinning knowledge (see Chapter 6) needs to be debated and addressed with non-technical information in policy decisions. The corporate model of health that was adopted by the states and territories in the 1990's allows corporations to fund health promotion programs, such as vaccination, to the public in media campaigns. If industry funding is involved in health promotion campaigns it is possible for vaccine manufacturers to influence public behavior through the selection or framing of information. The case study of the *Swine Flu Pandemic* in 2009 illustrates how the framing of disease statistics in the media can misrepresent risk and influence public behavior regarding the use of vaccines.

Surveillance statistics can be misrepresented to the public by changing the definition of the disease or changing the surveillance of the disease without publicising the changes. This information can then be promoted to the public in health promotion campaigns in the media that are funded by corporate partnerships. The media is not accountable for the information it provides to the public and the community is expected to *trust* that the information provided is in the public's best interest. This situation can compromise public health because industry promotes industry interests which are not always compatible with the public interest. Industry became more involved in health research and promotion in the latter part of the twentieth century with the development of the medical-industrial

partnership in institutions and public-private partnerships to sponsor health programs. These have been discussed in Part 2 of the thesis: Corporate Influence and Undone Science in Public Policy.

## **Chapter Overview**

Chapter 5 describes the scientific medical model of health and the ethical values inherent in health promotion and chapter 6 discusses the influence of corporations in the production of scientific knowledge. In chapter 7 I discuss the information the government provides to the public to support its claims about vaccines. Chapter 8 introduces the concept of undone science and the political framework that leads to its existence. Examples of undone science and industry influence in government vaccination polices are provided in case studies in chapters 9 and 10. Chapter 11 presents the conclusions drawn from this investigation.

**Chapter 5 Conclusion** 

**Chapter Title: Public Health Policy and Health Promotion Ethics** 

Chapter pages:118-136

#### 5.8 Conclusion

In the mid-20<sup>th</sup> century there was a change in focus in Australia's public health policies from the social and environmental determinants of health to the use of vaccines, a medical intervention, as the main focus for disease prevention. This was in line with progress in scientific (western) medicine that considers most causes of infectious disease to be biological and initiated from within the individual. Consequently it has a narrower focus on etiology that discounts the influence of environmental, political and social causes of disease. The central aim of the scientific medical model of health is to reduce morbidity and mortality in the population by increasing the use of medical interventions. This is evidenced in the development of Australia's vaccination policies which show an increase in the number of vaccines recommended by the government, and an increase in coercive measures, as the threat from infectious diseases declined. When powerful associations

shape medical knowledge they can influence the behaviour of individuals in society through public health policies. In Australia the medical profession and the pharmaceutical industry have a major role in determining areas of health policy with limited input from other interest groups, particularly the public on whom this policy is enforced. This situation can result in the appearance of a scientific consensus on vaccination because these professional bodies can choose the research topics that will be funded. When evidence is selected to support a desired outcome in a health policy the status quo can be maintained. The political framework that enables evidence to be selected for government policies is described in chapter 8. This outcome is also achieved by the professional medical associations through the regulation of doctors' education and by the standards set through the *Good Medical Practice guidelines*. Medical practitioners in Australia can be deregistered from the profession if they do not comply with the standards and regulations set by the Medical Board of Australia (MBA). This means doctors are not free to present their personal assessment of the risks and benefits of vaccination because their professional regulations require them to support government vaccination policies.

Vaccines were adopted as the primary preventative strategy for infectious diseases from the 1950's onwards. The sales of drugs and vaccines increased in the 1980's when the free market economic model was adopted in Australia even though the safety of many of these products had not been established in properly designed clinical trials. In this political model industry can provide financial incentives to doctors to increase the sales of their products. They can also provide sponsorship for doctors' education. When financial incentives are provided to enhance a doctor's livelihood the ethical guidelines for health promotion can be overlooked and this puts patient health at risk. Gaps may exist in the scientific knowledge that are essential to predicting health outcomes, yet these can be ignored if information is presented to doctors and patients in a selective manner or if conflicts of interest exist in decision-making boards. Doctors trust that they are getting a balanced education and in general patients trust the advice provided to them by doctors. Patients and doctors are dependent upon the information they receive to make an informed decision on a medical procedure. Consequently, maintaining patient autonomy and informed choice regarding medical interventions is fundamental to maintaining the health of populations. This is particularly important when the integrity of medical science, and sponsorship of education,

can be influenced by political and cultural factors. A policy that removes these values contravenes the guiding principles stated in the Seedhouse Ethical Grid for the promotion of health in the population.

Estimating the risk from an infectious agent depends upon the method of measuring disease burden in the community – case-fatality or incidence of the disease (see chapter 4). These methods are value-based and ethical guidelines imply that governments should seek the consent of the community before value-based interventions are implemented in public policy. When policies are presented in a benevolent manner, infringements on human rights can be overlooked if the community places unwarranted trust in the medical profession and the government. The public's interest in these policies should not be compromised by the economic or political interests of special groups. The Australian government has not provided adequate evidence for coercive measures in vaccination policies or that the recommended actions are proportionate to the risk of infectious diseases in Australia. These actions are also not included in any laws under Australian health acts. In addition, they are being implemented whilst the government states that vaccination in Australia is not compulsory. When certain political structures exist, governments can claim a procedure is safe simply because there is a *lack of evidence* for the public to prove otherwise. This is termed 'undone science' and the political structures that lead to areas of unfunded research relevant to public health policy are described in Part 2 of this thesis 'Corporate Influence and Undone Science in Public Policy'.

#### Part 2 Chapter Overview

The influence of corporations in the production and promotion of science is described in chapter 6 and a discussion of the government's claims about the safety and efficacy of vaccines is provided in chapter 7. In chapter 8 I describe the political framework that allows increasing areas of undone science to exist in the design of public health policy. Chapters 9 and 10 provide case studies of the HPV vaccine and the 'Swine Flu' 2009 vaccine to illustrate the influence of politics and corporations in the development of global and national vaccination policies. Chapter 11 provides the conclusions to this investigation.

**Chapter 6 Conclusion** 

**Chapter Title: Industry Influence in Research and Policy** 

Chapter pages:137-163

#### 6.9 Conclusion

In the 21<sup>st</sup> century universities and research institutions are operating in partnerships with industry and directing research into profitable technology. Universities receive large amounts of money from industry that are not transparent to the public. COI are ubiquitous in financial relationships involving researchers in university faculties. Consequently industry has unprecedented influence over the type of research that is performed and the outcomes achieved. COI also exist in relationships involving the medical profession, media and government. These relationships play a significant role in the way drugs/vaccines are promoted to the community. When industry funds the research it leads to less public

interest science being investigated because it might not serve industry interests. This is termed 'undone science' and the political framework for this practice is described in chapter 8. Vaccines/drugs are being approved for the market without properly designed clinical trials. The side-effects of drugs are being down-played to doctors and consumers and the benefits are over-emphasised. Many peer-review journals now depend upon industry funding for their profits and this increases the publication bias towards positive trial results and the suppression of negative results. Pharmaceutical companies are also sponsoring lobby groups that appear to be advocating for consumer interests but in fact are fronts for drug companies. This influence synchronises with pharmaceutical marketing to doctors which is presented as 'education' and the media promotion of vaccines influenced by corporations. Consequently there is a systematic bias towards industry interests in medical research and public health policy and promotion.

A lack of acknowledgment by governments of an important area of research is easier to maintain if the stakeholder whose interests are affected is removed from the political decision-making process. This is observed in the development of Australia's vaccination policies as the community is not consulted or encouraged to participate in public debate on vaccination and there is only one consumer representative on the government vaccine advisory committee (ATAGI). In addition, pharmaceutical representatives can be invited to ATAGI committee meetings to provide information but these meetings are not open to the public and the information is not available for public scrutiny before vaccines are approved. A lack of political power and financial support also has the effect of reducing the consumer voice in the mainstream media. These factors are synchronising to remove an independent consumer perspective from the risk assessment process of policy development. They are also resulting in non-transparent policy decisions being made by ATAGI/NCIRS members in an unsystematic assessment of the risks.

The lack of independent regulation of the global vaccine market is resulting in sub-standard vaccines. Vaccines are a global production and they can be automatically approved in many countries based on clinical trials that were performed in another country. Manufacturers in the US have less incentive to develop safe and effective vaccines because they are exempt from liability when harm is caused. This legislation ensures that there is a stable vaccine

market but it does not provide incentives to protect the health of the population. Government regulators in most countries are 100% funded by industry under a Cost-Recovery (User-Pay) system. This means they approve their sponsor's vaccines/drugs for the market and monitor these same products for safety and efficacy. In effect they are indirectly monitoring their own products. Large political donations from pharmaceutical companies are also being allowed to influence government policy. Funded lobby groups are targeting policy decision-makers, medical practitioners, educational boards and mainstream media with selective information. Vaccine advisory boards are rife with conflicts of interest, enabling industry to influence the direction of government funding in health policy research and policy decision-making. National vaccine advisory committees such as ATAGI have been established in many WHO member countries and they receive advice and financial support from the WHO in the development of national vaccination programs. Recommendations for new vaccines are not always founded on local data and costeffectiveness is being determined using economic models that rely on non-transparent assumptions about the safety and efficacy of vaccines. Although the importance of vaccinecreated herd immunity is used to promote vaccines to the community, the chairman of ATAGI for the last decade states that the implications for herd immunity for new vaccines are 'neither necessary nor sufficient for a positive recommendation for NIP suitability' (Nolan 2010 A79). This indicates that vaccines are being promoted to the community on a false premise that has serious implications for population health. Further, the costeffectiveness of vaccines is being determined on evidence produced in clinical trials that are funded by pharmaceutical companies and carried out by researchers/chief investigators who are representatives on government vaccine advisory boards such as ATAGI and the NCIRS.

This arrangement is very profitable for universities, governments, researchers and representatives on vaccine advisory boards but it is extremely costly to public taxpayers and to population health. In 2008-2009 the cost of providing vaccines 'free' to Australians was well above \$AU400 million (Nolan 2010 A76). However, the actual cost of these programs is unknown because the figures are not released to the public (even when requested) and they do not include the cost to the community of the deaths and disability that are a known side-effect of vaccines. This cost to the community is unknown because

the TGA has not established an active surveillance system that can make causal relationships to vaccines. A regulator that is 100% funded by industry has no incentive to accurately monitor the adverse events from its own products. This demonstrates the need for vaccination policies to be independent from commercial and political interference in order to protect public health. In Australia policy decisions for vaccination programs are based on research (often unpublished) that is performed by government representatives on vaccine advisory boards who receive honoraria and funding for their clinical trials from pharmaceutical companies. The findings from such research are being used in policy decisions for vaccination programs without public scrutiny or assessment by independent researchers.

In chapter 7 I discuss the evidence the Australian government is providing to the public to support the claims about vaccine safety and efficacy. Chapter 8 presents a description of undone science and the political framework that leads to a lack of integrity and rigour in medical science. Chapters 9 and 10 are case studies of the HPV vaccine and 'Swine Flu' 2009 vaccine, showing the influence of corporations in the development of global vaccination policies. Chapter 11 presents the conclusions for this investigation.

# **Chapter 7 Conclusion**

Chapter Title: The Evidence Underpinning Claims about Vaccines

Chapter pages: 164-194

#### 7.7 Conclusion

The evidence provided by the Australian government and the AAS for vaccinating with multiple vaccines does not include an assessment of the ecological complexity of the cause of infectious diseases or account for the genetic diversity of the population. It also does not provide direct evidence of the influence of vaccines in controlling any infectious diseases. In addition, the adverse events from using multiple vaccines in infants/adults should be considered in the adoption of a management strategy that is implemented in public health policy. The government information analysed here does not provide estimates of the frequency and type of risk associated with each vaccine - or with the combination of vaccines. It also does not provide evidence that policy-decisions about infectious diseases are being made for the benefit of the majority of the community. The value judgments made by policy advisors in Australia are emphasising the assumed benefits of vaccines and downplaying the risks. This is illustrated in the selective evidence and misleading statements that are used to promote vaccines to the public in the FAQ and by the AAS. Conflated terminology has been used to mislead the public about the efficacy of vaccines. The government has not provided evidence that the 'best judgments' for public policy are being made on comprehensive and independent evidence. Australia's vaccination policies include undone research and a lack of transparency in the rigour of scientific trials and the assumptions used in the evaluation of vaccines. This is a consequence of a culture that promotes scientific research for 'profit' as opposed to its contribution to progressing knowledge. In the 21<sup>st</sup> century industry is sponsoring vaccine clinical trials without evaluation from independent experts. This is not disinterested science and it is being promoted in public policy by experts with vested interests.

The culture in which research and policy development is occurring in Australia is described in chapters 6 and 8. An example of the influence of corporations in global vaccination policies is provided with the HPV vaccine in chapter 9 and the 'Swine Flu' 2009 vaccine in chapter 10. These case studies demonstrate how vaccines can be implemented into global vaccination polices even though the underpinning science is incomplete. Chapter 11 presents the conclusions to this investigation.

# **Chapter 8 Conclusion**

# Chapter Title: Politics and Undone Science in Public Policy

Chapter pages: 195-216

#### 8.11 Conclusion

Undone science is the research that is not carried out because the likely results would be unwelcome to powerful groups. When areas of science are not funded, policies can be designed on incomplete knowledge and result in unpredictable health outcomes in the population. This is because there are gaps in the scientific knowledge that underpins the value judgments that are made in policy decisions. Government policies that are designed on incomplete scientific knowledge may not protect the public interest. Areas of undone science have expanded in the era of globalization because industry is influencing the research agendas of both governments and research institutions. This is a consequence of industry being in partnership with universities and being the main sponsor for many private research institutions. In addition, government funding agendas have aligned with industry goals because the dominant network of scientists is influential on government advisory boards. A shift to industrial sources of funding for scientific research is also achieved when there is increased scope for the patenting and licensing of scientific discoveries. These developments lead to increased conflicts of interest in the production of scientific knowledge and in the design of public health policy, along with a lack of transparency. They also have the effect of reducing government emphasis on public interest research and increasing the pockets of undone science in different fields of science.

In Australia the federal health minister receives advice on vaccination policies from technical experts without consultation with the community. The institutional barriers that exist in the Australian political system and in academic and media organizations have resulted in the marginalisation of the public's contribution to vaccination policy, even though the public is a main stakeholder in these policies. The Australian government justifies current vaccination policies by claiming, in effect, there is no evidence of harm so therefore no action is required. However, this claim ignores the lack of evidence due to research that has not been funded. The undone science in Australia's vaccination policies includes comprehensive RCT's for the efficacy and safety of vaccines, either singly or in the combined schedule of vaccines in animals and humans. In this situation it becomes necessary for the public to have influence in the decision-making process to ensure that all relevant research is undertaken and accounted for in policy decisions.

The lack of integrity and rigour in the clinical trials and scientific research that is being sponsored by industry for global health policies was discussed in chapters 6 and 7. In chapters 9 and 10, I have provided case studies illustrating the undone science and industry influence in the promotion of global HPV vaccination programs and the pandemic 'Swine Flu' 2009 vaccine. These case studies illustrate how vaccines can be developed and policies implemented on incomplete science due to the political framework that results in undone science.

**Chapter 9 Conclusion** 

Chapter Title: Case Study: The Human Papillomavirus Vaccine (HPV)

Part 1 HPV and Cervical Cancer Pathogenesis

Chapter pages: 217-257

#### 9.15 Conclusion to Part 1

The promotional campaigns for HPV vaccine misrepresented the risk of HPV infections and cervical cancer to women in different countries. This was done in order to create a market for the vaccine. Currently the benefit of HPV vaccines against the burden of cervical cancer is unknown and the risk of injury or death associated with the vaccines has not been accurately determined in different populations. Universal HPV vaccination programs that target infection from HPV 16 and 18 are not beneficial to the majority of women in developed countries because these infections will not progress to cancer without specific environmental and genetic factors (risk factors) also being present. HPV infections do not progress to carcinoma in the majority of cases and there are currently 13+ oncogenic strains of HPV that the vaccine does not protect against.

HPV vaccines are not demonstrated to be safer or more effective than Pap screening combined with surgical procedures. Hence it follows that implementing broad HPV vaccination programs is not necessary or cost-effective because Pap screening programs

combined with surgery are the most effective prevention and are still required by vaccinated women. HPV vaccines are offering uncertain benefits in reducing the burden of cervical cancer and may cause more harm than good due to the lack of investigation of their long-term safety.

**Chapter 9 Conclusion** 

**Chapter Title: Case Study: The Human Papillomavirus Vaccine (HPV)** 

Part 2 Undone Science in HPV Vaccination Programs

Chapter pages: 258-268

#### 9.20 Conclusion to Part 2

It is important that comprehensive research on the safety and efficacy of drugs is completed prior to their implementation. When crucial research is left unfunded, policy decisions are founded on incomplete evidence (selective science). In addition, it is essential that any research that is performed and funded by industry is open to assessment by independent scientists before the conclusions are accepted in public health policies. Research that is performed by scientists with ties to industry may result

in biased results and this has significant implications for population health when the findings are recommended to policy advisors by representatives who also have financial ties to industry.

Hidden COI in all areas of research and policy development are institutional barriers that prevent the government from being accountable to the public. This is because industry agendas influence the research and industry employees double as representatives on government and professional decision-making bodies. Ultimately, the price that is paid for adopting private-public sponsorship of global health policies and the academic-industry model for research institutions is an increased risk to the health of populations because the profit motive contaminates the search for knowledge. This health model is a faith-based system and not an evidence-based system because it is dependent upon trade secrets and political decisions that are based on biases that are not transparent to the public.

**Chapter 10 Conclusion** 

Chapter Title: Case Study: 'Swine Flu' 2009 Pandemic

Chapter pages: 269-93

#### 10.13 Conclusion

The 'Swine Flu' pandemic of 2009 was declared by a secret WHO committee that had ties to pharmaceutical companies that stood to make excessive profits from the pandemic. This situation was facilitated by the lack of effective regulations and transparency regarding COI, within the WHO and national governments, to prevent pharmaceutical companies from exploiting global health policies to their advantage. The pattern of this pandemic was similar to that of the 'swine' flu pandemic of 1976 which also did not eventuate. Scientists are using industry funded research to predict the occurrence of influenza pandemics and these are based upon questionable premises and assumptions linked to the profits that can be made if a pandemic is declared. Under the influence of GAVI significant changes were made to the governance of global health policies through the implementation of the International Health Regulations (IHR) in 2005. These changes required all WHO member countries to increase surveillance systems for all infectious diseases to create extensive national surveillance systems: not just at the entry and exit points for countries. GAVI/WHO also spent 10 years developing Pandemic Preparedness Plans (PPP) with all member countries to ensure that if a global pandemic was declared all countries would be required to buy vaccines at prices that were set in 'silent contracts'. By 2005 nascent technologies had enhanced the surveillance of diseases by enabling greater sub-typing of infectious agents. Whilst increased surveillance of a disease can give the appearance of a greater risk due to more cases, this is not always the case. For many diseases, such as influenza, the majority of cases are mild and without complications, and would otherwise have gone unnoticed. This risk assessment of disease incidence and severity needs to be characterised for each disease because it is the severity of a disease and not the incidence that presents a risk to population health (Burnet 1956; Cumpston 1989). This characteristic of risk assessment for infectious diseases was discussed in chapter 4.

Influenza is a common respiratory virus that mutates regularly. Whilst a new strain of virus *may* be more virulent it is important to establish the risk prior to introducing a new vaccine because vaccines themselves produce a risk to individuals and the efficacy and safety of influenza vaccines are still being debated. The interpretation of the PP used in government policies is a political decision that is linked to the desired outcomes of the policy. An interpretation that does

not put the onus of proof on the proponent does not protect the public interest in health policies. The information about the risks of vaccines is significant to public health but it is not communicated to the public by governments or the mass media. In the case of the 2009 'Swine Flu' pandemic, a level 6 pandemic could not have been declared if the WHO secret Emergency Committee had not been given the power to change the definition of a pandemic. This power was given to the EC by the European Scientific Working Group on Influenza that was 100% funded by industry. Public health is at risk if authorities and the media are not accurately informing the public about the risks of both diseases and vaccines. For many diseases increased surveillance results in a large number of identified cases that do not present a greater threat to public health. This was the case for the increased surveillance of 'Swine Flu' in 2009 and it resulted in an excessive waste of government funds and endangered public health.

**Chapter 10 Conclusion Chapter Title: Conclusion** 

Chapter pages: 294-307

#### 11.4 CONCLUSION

In chapter 1 I stated that the aim of this thesis is to examine the following claims made by the Australian government (section 1.3):

- 1. Vaccines are *proven* to be a safe, effective and necessary management strategy for infectious diseases and
- 2. The benefits of vaccines to the community far *outweigh* the risks of vaccines to individuals and population health.

This thesis has demonstrated that there is inadequate evidence from independent studies to support these claims about vaccines.

Claim 1: Most mass vaccination campaigns were introduced into developed countries after 1950 in an attempt to *eliminate* infectious diseases, not because infectious diseases were a serious risk to the majority of Australian children. In addition, many vaccines were introduced after 1980 on a directive from the WHO at a time when infant mortality rates and the risk of infectious diseases in Australia were very low. There is a lack of evidence to support the claim that vaccine-induced herd immunity can eliminate infectious diseases. The Australian government has not provided evidence from formal controlled clinical trials that demonstrate the efficacy of vaccines in preventing disease nor has it provided transparent, independent data on the vaccination status, socioeconomic status, and severity of reported cases of disease that would demonstrate the influence of vaccines in preventing disease.

There is a lack of evidence to claim that vaccines are a safe and effective management strategy in diverse genetic populations because the appropriate scientific studies have not been funded to determine the types and frequency of adverse events that are occurring in different communities.

Claim 2: The government's claim that the benefits of vaccines to the community far outweigh the risks of vaccines to individuals cannot be sustained because the government has

not established an active surveillance system that can provide data on the long-term health effects of single or multiple vaccines in the Australian population. The correlation between the increased use of vaccines in the NIP and the significant increase in chronic illness in children has not been acknowledged or investigated by the Australian government. Furthermore, the Australian government claims that the benefit of using vaccines is to create herd immunity to protect the community. Yet the criterion for recommending a vaccine for approval in the NIP does not include the necessity to demonstrate that the vaccine has implications for herd immunity in the population (Nolan 2010 A79).

#### **Summary:**

The undone science in Australian government vaccination policies results in unpredictable health outcomes in the population. It needs to be acknowledged by the Australian government that increased morbidity in the Australian population is a possible outcome if vaccination policies are being implemented on a lack of comprehensive scientific evidence. Infectious diseases had declined in severity in Australia before most vaccines were introduced. Vaccines were not introduced to reduce the deaths and illness due to infectious diseases but to see if they could be eliminated. Infectious diseases were re-labeled vaccine-preventable diseases to imply that vaccines are the key to preventing disease, which contradicts the historical decline of infectious diseases. The contribution that vaccines may have made to the decline of infectious diseases is unknown because there has been no systematic assessment of their efficacy in preventing disease. Australian government vaccination policies have been founded on global directives from the WHO/GAVI that were designed by an alliance with industry. They have not been recommended as a result of an independent assessment of the need for vaccines in the ecological context of Australia. This is significant because environmental and host characteristics play a role in disease expression. Infectious agents on their own do not cause disease and this explains why many infectious diseases are not severe in countries with improved environmental and social conditions. The Australian government has not provided a risk/benefit assessment for each vaccine in the Australian context to support the claims that are made in vaccination policies.

Finally it must also be acknowledged that industry influence is pervasive and is affecting research findings as well as the research topics that are being investigated and the decisions being made in policy. Undone research in the government's vaccination policies includes the absence of studies of vaccination by scientists who are completely independent of industry

influence. In addition, decisions are being made by policy decision-makers with COI with industry that are not transparent to the Australian public, even if they are required to be declared before a meeting. The public is also not informed that directives for Australia's public health policies are provided by the WHO/GAVI alliance that includes partnerships with pharmaceutical companies. This synchronises with government vaccine advisory boards that do not have adequate representation of public members to present the public perspective in these policies. In this situation, the public is left to trust that officials are using evidence that is produced with integrity and rigour to develop policies that protect the public interest. Biomedical research produced in academic/industry research institutions contains trade secrets that prevent collaboration and an independent assessment of the science. There is also a lack of independent studies to assess the accuracy of the conclusions that are drawn. This leads to institutional biases in the political decisions that are made in national and global public health policies. Government policies are being promoted to the public in the interests of the community yet they are being designed and driven by industry interests. This misleads the public and endangers public health. Governments are also informing the public that they should seek their advice on vaccination from medical practitioners. Yet health professionals are unable to speak freely about the risks and benefits of vaccines because they are required to support government vaccination polices for their professional registration.

These conclusions emphasise the need for maintaining voluntary participation, without coercion, in public health policies that include a medical intervention. This is particularly the case for preventative health policies that promote a medical intervention to *healthy* individuals. Healthy communities are achieved by increasing individual autonomy, that is, the individual's right to choose how they care for their own bodies in the prevention of disease. This prevents indoctrination and it must be respected and promoted in public health policies to ensure that better health is the primary outcome of these policies.

## **Bibliography**

Ada G and Isaacs D. 2000. *Vaccination: the facts, the fears, the future*. Sydney: Allen and Unwin.

Allen A. 2007. Vaccine: The Controversial Story of Medicine's Greatest Livesaver. New York: Norton.

Angell M. 2005. The Truth About Drug Companies: How they deceive us and what to do about it. New York: Random House.

Angell M. 2009. Drug Companies and Doctors: a Story of Corruption, *The New York Review of Books*. <a href="http://www.nybooks.com/articles/archives/2009/jan/15/drug-companies-doctorsa-story-of-corruption/">http://www.nybooks.com/articles/archives/2009/jan/15/drug-companies-doctorsa-story-of-corruption/</a> (accessed Nov 2011).

Archibald K, Coleman R, Foster C et al. 2011. Open Letter to UK Prime Minister David Cameron and Health Secretary Andrew Lansley on Safety of Medicines. *The Lancet*. 377: 9781 (June): p1915.

Armstrong G, Conn LA, Pinner RW. 1999. Trends in Infectious Disease Mortality in the United States During the 20<sup>th</sup> Century. *Journal of the American Medical Association*. 281: 1 (Jan): pp61-66.

Association of Faculties of Medicine of Canada (AFMC). 2007. Primer on Population Health. A virtual textbook on Public Health concepts for clinicians. Determinants of Health and Health Inequalities, Basic Concepts: causes, risks, factors and determinants. The Interacting Triad of Causal Factors. AFMC Public Health Educators' Network <a href="http://phprimer.afmc.ca/">http://phprimer.afmc.ca/</a> (accessed October 2013). License: Creative Commons BY-NC-SA

Austin DW, Shandley KA, Palomba EA. 2010. Mercury in vaccines from the Australian childhood immunisation program schedule. *Journal of Toxicology and Environmental Health -Part A: Current Issues.* 73: 10: pp637-640.

Australasian Society of Clinical Immunology and Allergy (ASCIA). 2015. First National Allergy Strategy released. Press release. 7 August <a href="http://www.allergy.org.au/about-ascia/media/617-aug-7-2015-first-national-allergy-strategy-released">http://www.allergy.org.au/about-ascia/media/617-aug-7-2015-first-national-allergy-strategy-released</a> (accessed August 2015).

Australian Academy of Science (AAS). 2012. The Science of Immunisation: Questions and Answers. http://www.science.org.au/policy/immunisation.html (accessed December 2012)

Australian Government (AG). Department of Health and Ageing.

- i) Aged Care Australia, Influenza Fact Sheet (accessed September 2011)
- ii) About Pandemic Influenza (accessed September 2011)
- iii) Office of Health Protection (OHP) 2009

Australian Government (AG). Department of Health and Ageing.

a) Communicable Diseases Control. 2007. Vaccine Preventable Diseases and Vaccination Coverage in Australia 2003-2005, Appendix 5: Government funding of National Immunisation Programs in Australia. *Communicable Diseases Intelligence Journal (CDIJ)*, Vol 31 – Supplement. June.

- http://www.health.gov.au/internet/publications/publishing.nsf/Content/cda-cdi31suppl.htm~cda-cdi31suppl-apx5.htm (accessed May 2013).
- b) enHealth Council 2002. Environmental Health Risk Assessment (EHRA) Guidelines for assessing human health risks from environmental hazards. <a href="http://www.health.gov.au/internet/nhhrc/publishing.nsf/Content/132/\$FILE/132%20Noise%20Watch%20Australia%20Submission%20Attachment%20B.pdf">http://www.health.gov.au/internet/nhhrc/publishing.nsf/Content/132/\$FILE/132%20Noise%20Watch%20Australia%20Submission%20Attachment%20B.pdf</a> (accessed October 2013).
- c) Immunise Australia Program (IAP) <a href="http://www.immunise.health.gov.au">http://www.immunise.health.gov.au</a> (Accessed 2006 and 2012, 2013).
- d) Immunise Australia Program (IAP): HPV School Vaccination Program http://hpv.health.gov.au/ (accessed June 2014)
- e) Immunise Australia Program (IAP) Advisory Boards. <a href="http://www.health.gov.au/internet/immunise/publishing.nsf/content/advisory-bodies">http://www.health.gov.au/internet/immunise/publishing.nsf/content/advisory-bodies</a> (accessed September 2011).
- f) Immunise Australia Program (IAP) Frequently Asked Questions. <a href="http://immunise.health.gov.au/internet/immunise/publishing.nsf/Content/faq">http://immunise.health.gov.au/internet/immunise/publishing.nsf/Content/faq</a> updated July 2013 (accessed 2013).
- g) Immunise Australia Program (IAP). Immunisation Related Payments for Parents, Strengthening Immunisation for Children. <a href="http://immunise.health.gov.au/internet/immunise/publishing.nsf/Content/factsheet-strengthening-immunisation">http://immunise.health.gov.au/internet/immunise/publishing.nsf/Content/factsheet-strengthening-immunise.health.gov.au/internet/immunise/publishing.nsf/Content/related-payments</a> updated September 2013 (accessed Nov 2013).
- h) Immunise Australia Program (IAP). Frequently Asked Questions: Changes to National Immunisation Schedule (CNIS) and related payments. <a href="http://immunise.health.gov.au/internet/immunise/publishing.nsf/Content/faq-related-payments">http://immunise.health.gov.au/internet/immunise/publishing.nsf/Content/faq-related-payments</a> updated October 2013 (accessed Nov 2013).
- i) Immunise Australia Program (IAP). National Immunisation Program Schedule of Vaccines.
   <a href="http://immunise.health.gov.au/internet/immunise/publishing.nsf/Content/faq-related-payments#immunised">http://immunise.health.gov.au/internet/immunise/publishing.nsf/Content/faq-related-payments#immunised</a> updated February 2013 (accessed October 2013)
- j) Immunise Australia Program (IAP). Australian Technical Advisory Group on Immunisation (ATAGI) <a href="http://www.immunise.health.gov.au/internet/immunise/publishing.nsf/Content/FC7BB2DC63225F8ACA257D770012DBF7/\$File/2015-ATAGI-conflict-interest.pdf">http://www.immunise.health.gov.au/internet/immunise/publishing.nsf/Content/FC7BB2DC63225F8ACA257D770012DBF7/\$File/2015-ATAGI-conflict-interest.pdf</a> (accessed August 2015).
- k) National Centre for Immunisation, Research and Surveillance (NCIRS). School Entry Vaccination Requirements: Summary of the Evidence. <a href="http://www.immunise.health.gov.au/internet/immunise/publishing.nsf/Content/ncirs-sevr-se">http://www.immunise.health.gov.au/internet/immunise/publishing.nsf/Content/ncirs-sevr-se</a> updated June 2013 (accessed November 2013).
- l) National Centre for Immunisation, Research and Surveillance (NCIRS). 2009. Vaccine Components. <a href="http://www.ncirs.edu.au/immunisation/fact-sheets/vaccine-components-fact-sheet.pdf">http://www.ncirs.edu.au/immunisation/fact-sheets/vaccine-components-fact-sheet.pdf</a> (accessed 2012).
- m) National Centre for Immunisation Research and Surveillance (NCIRS). 2009. Thiomersal. <a href="http://www.ncirs.edu.au/immunisation/fact-sheets/thiomersal-fact-sheet.pdf">http://www.ncirs.edu.au/immunisation/fact-sheets/thiomersal-fact-sheet.pdf</a> (accessed 2012).
- n) National Centre for Immunisation, Research and Surveillance (NCIRS). Executive. <a href="http://www.ncirs.edu.au/staff/executive/index.php">http://www.ncirs.edu.au/staff/executive/index.php</a> (accessed August 2015).

- o) Overview of Environmental Health (OEH). <a href="http://www.health.gov.au/internet/main/publishing.nsf/Content/health-pubhlth-strateg-envhlth-index.htm">http://www.health.gov.au/internet/main/publishing.nsf/Content/health-pubhlth-strateg-envhlth-index.htm</a> (accessed October 2013).
- p) The Australian Immunisation Handbook. 9<sup>th</sup> Edition. 2012. Components of Vaccines. Appendix 4. Australian Technical Advisory Group on Immunisation (ATAGI). <a href="http://www.immunise.health.gov.au/internet/immunise/publishing.nsf/Content/78">http://www.immunise.health.gov.au/internet/immunise/publishing.nsf/Content/78</a> CDF41C283426A8CA2574E40020CCAB/\$File/handbook-9.pdf (accessed March 2013).
- q) *The Australian Immunisation Handbook.* 10<sup>th</sup> Edition. 2013. Australian Technical Advisory Group on Immunisation (ATAGI). <a href="http://www.immunise.health.gov.au/internet/immunise/publishing.nsf/Content/Handbook10-home">http://www.immunise.health.gov.au/internet/immunise/publishing.nsf/Content/Handbook10-home</a> (Updated 30<sup>th</sup> May 2013).
- r) The Australian Immunisation Handbook. 10<sup>th</sup> Edition. 2013. Appendix 3: Components of vaccines used in the National Immunisation Schedule (NIP). Australian Technical Advisory Group on Immunisation (ATAGI). <a href="http://www.immunise.health.gov.au/internet/immunise/publishing.nsf/Content/appendix3">http://www.immunise.health.gov.au/internet/immunise/publishing.nsf/Content/appendix3</a> (accessed October 2013).
- s) Therapeutic Goods Administration (TGA). Reporting Medicine and Vaccine Adverse Events. <a href="http://www.tga.gov.au/hp/problem-medicine.htm">http://www.tga.gov.au/hp/problem-medicine.htm</a> (accessed March 2013).
- t) Therapeutic Goods Administration (TGA). 2009. Cost Recovery Impact Statement (CRIS). Regulation of In-Vitro Diagnostic Devices. September. <a href="http://www.tga.gov.au/archive/fees-cris-ivd-091020.htm#n2">http://www.tga.gov.au/archive/fees-cris-ivd-091020.htm#n2</a> (accessed June 2012)
- u) Therapeutic Goods Administration (TGA). 2012. Cost Recovery Impact Statement (CRIS). July 2012 2013 <a href="http://www.tga.gov.au/pdf/fees-cris-gmp-120629.pdf">http://www.tga.gov.au/pdf/fees-cris-gmp-120629.pdf</a> (accessed June 2013).
- v) Therapeutic Goods Administration (TGA). Industry Working Group (IWG). <a href="http://www.tga.gov.au/about/tga-bpi-pm-bpr-iwg.htm#meetings">http://www.tga.gov.au/about/tga-bpi-pm-bpr-iwg.htm#meetings</a> (accessed June 2013).
- w) Therapeutic Goods Administration (TGA). 2010. Response to the West Australian (Stokes) Review (RWAR) into the handling of AEFIs following 2010 seasonal flu vaccination. November. <a href="http://www.tga.gov.au/safety/alerts-medicine-seasonal-flu-101018.htm">http://www.tga.gov.au/safety/alerts-medicine-seasonal-flu-101018.htm</a> (accessed 2011).

Australian Government. 2015. Department of Health (DH). Copyright, Linking, Disclaimer and Privacy. April

http://immunise.health.gov.au/internet/immunise/publishing.nsf/Content/copyright#disclaime r (accessed August 2015).

Australian Government. 2015. Department of Human Services. Payment Rates for Childcare Benefit. June. <a href="http://www.humanservices.gov.au/customer/enablers/centrelink/child-care-benefit/payment-rates">http://www.humanservices.gov.au/customer/enablers/centrelink/child-care-benefit/payment-rates</a> (accessed July 2015).

Australian Government. 2015. Australian Parliament (AP). Twenty-second Report of the 4<sup>th</sup> Parliament: a) Chapter 1 Social Services Legislation Amendment Bill 2015 (May): p105 b) Chapter 1 Right to Privacy: p58.

Australian Government. 2008. Department of Industry and Innovation. The Prime Minister's Prizes for Science. Ian Frazer: A hero of women and science.

https://grants.innovation.gov.au/SciencePrize/Pages/Doc.aspx?name=previous\_winners/PM2 008Frazer.htm (accessed September 2013).

Australian Institute of Health and Welfare (AIHW). 2005. Child health, development and wellbeing. Australian Government:

- a) Selected Chronic Diseases Among Australia's Children. Bulletin 29. September 2005.
- b) Chronic Diseases and Associated Risk Factors.
- c) A Picture of Australia's Children. May. 2005 (accessed March 2006).

Australian Institute of Health and Welfare (AIHW). 2006. Facts about Cervical Cancer. Australian Government.

Australian Medical Association (AMA). 2006. Declaration of Geneva. Guiding principles of a Medical Practitioner. http://ama.com.au/node/2468 (accessed Nov 2011).

Autism Spectrum Australia. Autism Association of NSW (Aspect) <u>www.autismnsw.com.au</u> (accessed April 2006).

Bale C, Garly ML, Martins C, Nielson J, Whittle H, Aaby P. 2011. Risk factors for measles in young infants in an urban African area with high measles vaccination coverage. *Pediatric Infectious Disease Journal*. 30: 8 (Aug): pp689-693.

Barzley M. 1992. *Breaking through Bureaucracy: A New Vision for Managing Government*. Los Angeles: University of California Press. Cited in Stoneham et al. 2004.

Basch PF. 1994. *Vaccines and World Health. Science, Policy and Practice*. New York and Oxford: Oxford University Press.

Bashford A. 2002. Foreign bodies: vaccination, contagion and colonialism in the nineteenth century. In Bashford A and Hooker C. (eds). *Contagion: epidemics, history and culture from smallpox to anthrax*. South Australia: Hyde Park Press.

Bashford A and Hooker C. 2002. Introduction; contagion, modernity and post modernity. In Bashford A and Hooker C. (eds). *Contagion: epidemics, history and culture from smallpox to anthrax*. South Australia: Hyde Park Press.

Baum F. 2008. The New Public Health. South Melbourne: Oxford University Press.

Baxter K. 2010. Structural Barriers to Reform of the Australian Health and public hospital system. Australian Centre for Health Research (ACHR). Canberra: TFG International Pty Ltd. January.

Behague D, Tawiah C, Rosato M, Some T, Morrison J. 2009. Evidence-based policy-making: The implications of globally-applicable research for context-specific problem-solving in developing countries. *Social Science and Medicine*. 69: 10: pp 1539-1546.

Behrman RE and Kliegman R.M. eds. 1998. (third ed.). *Nelson Essentials of Pediatrics*. Philadelphia: WB Saunders Company.

Bennett C. 2012. Know Your Product: complementary medicines. Radio National Life Matters. Australian Broadcasting Commission (*ABC*). Australian Health Consumers Forum (HCF) <a href="http://www.abc.net.au/radionational/programs/lifematters/know-your-product3a-complementary-medicines/4379630">http://www.abc.net.au/radionational/programs/lifematters/know-your-product3a-complementary-medicines/4379630</a> (accessed Nov 2012).

Bennett C, Kuhn A, Haverkos H. 2010. Human papillomavirus and tar hypothesis for squamous cell cervical cancer. *Journal of Biosciences*. 35: 3 (September): pp331-337.

Bita N. 2010. "A flu jab too close for comfort". *The Australian*. 29<sup>th</sup> September.

Bita N. 2011. "Flu Advisor Defends CSL Shares." The Australian, 8<sup>th</sup> September.

Blaxill M and Olmsted D. 2013. A License to Kill? In Habakus LK and Holland M with Rosenberg KM (eds). *Vaccine Epidemic: how corporate greed, biased science and coercive government threaten our human rights, our health and our children*. New York: Skyhorse Publishing. pp175-185.

Blumberg DA, Lewis K, Mink CM, Christenson PD, Chatfield P, Cherry JD. 1993. Severe reactions associated with diphtheria—tetanus-pertussis vaccine: detailed study of children with seizures, hypotonic-hyporesponsive episodes, high fevers, and persistent crying. *Pediatrics*: 91: pp1158-1165.

Blume S, Jani J and Roalkvim S. 2013a. Saving children's lives: perspectives on immunisation. In Roalkvam S, McNeill D and Blume S (eds). *Protecting the World's Children: Immunisation Policies and Practice*. Oxford: Oxford University Press. pp1-30.

Blume S, Roalkvim S and McNeill D. 2013b. Concepts and Approaches. In Roalkvam S, McNeill D and Blume S (eds). *Protecting the World's Children: Immunisation Policies and Practice*. Oxford: Oxford University Press. pp31-58.

Bosch FX, Manos M, Munoz N, Sherman M, Jansen A, Peto J, Schiffman M, Moreno V, Kurman R, Shah K. 1995. International Biological Study on Cervical Cancer (IBSCC). Prevalence of Human Papillomavirus in Cervical Cancer: a Worldwide Perspective. *Journal of the National Cancer Institute*. 87: 11 (June): pp796-802.

Bosch FX, Lorincz A, Munoz N, Meijer CJLM, Shah KV. 2002. The causal relation between human papillomavirus and cervical cancer. *Journal of Clinical Pathology*. 55 (Jan): pp244-265.

Bosch FX, Broker TR, Forman D, Moscicki A, Gillison ML, Doorbar J, Stern PL, Stanley M, Arbyni M, Poljak M, Cuzick J, Castle PE, Schiller JT, Markowitz LE, Fisher WA, Canfell K, Denny LA, EL Franco, Steben M, Kane MA, Schiffman M, Meijer CJLM, Sankaranarayanan R, Castellsaguéa X, Kim JK, Brotons M, Alemanya L, Alberoa G, Diaz M, de Sanjosé S, on behalf of the authors of the ICO Monograph. 2013. Comprehensive Control of HPV Infections and Related Diseases. *Vaccine*. 31(S8): ppH1-H31.

Bosely S. 2002. Scandal of Scientists Who Take Money for Papers Ghostwritten by Drug Companies. *The Guardian Weekly*. February 7.

Boulianne N, De Serres, GDuval B, Joly JR, Meyer F, Dery P, Alary M, Le Henaff D, Theriault N. 1991. Major Measles epidemic in the region of Quebec despite 99% vaccine coverage. *Canadian Journal of Public Health*. 82: pp189-190.

Bradley A. 2013. GP's unwilling to sign vax refuser forms. *Australian Doctor.com*. June. <a href="http://www.australiandoctor.com.au/news/latest-news/gpsunwillingtosignvaxrefusorforms">http://www.australiandoctor.com.au/news/latest-news/gpsunwillingtosignvaxrefusorforms</a> (accessed September 2013).

British Society for Ecological Medicine (BSEM). 2009. Ecological Medicine. June 7. http://www.ecomed.org.uk/about-the-society/ecological-medicine (accessed Nov 2011)

Bryson M, Duclos P, Jolly A, Cakmak N. 2010. A global look at National Immunisation Advisory Groups. *Vaccine*. 28 (April) Suppl 1: A13-A17.

Bunker JP. 2001. The role of medical care in contributing to health improvements within societies. *International Journal of Epidemiology*. 30: pp1260–1263.

Burgess MA, McIntyre PB, Heath TC. 1998. Pertussis re-emerging: Who is responsible? *Australian Journal of Public Health*. 22: 1 (Feb): pp9-10.

Burnet FM. 1952. The Pattern of Disease in Childhood. *Australasian Annals of Medicine*. 1: 2: pp93-107.

Burton D. 2003. Mercury in Medicine Report, US Congressional Record; Findings and Recommendations, Safe Exposure Standard as Reported in Executive Summary. 20<sup>th</sup> May. www.aapsonline.org/vaccines/mercinmed.pdf (accessed June 2013).

Buse K and Waxman A, 2001, Public-private health partnerships: a strategy for WHO. *Bulletin of the World Health Organisation*. 79 (January): 8

Cameron S. 2004. Communicable Disease Control. Ch.13. In Cromar N, Cameron S and Fallowfield H. (eds.) *Environmental Health in Australia and New Zealand*. Melbourne: Oxford University Press. pp198-212.

Canadian Food Inspection Agency (CFIA). 2009. An Alberta Swine Herd Investigated for H1N1 flu virus. News Release. Media relations: 613-773-6600 May.

Carter W and Bowen J. 1991. *The Macquarie Home Guide to Health and Medicine*. The Macquarie Library Pty Ltd.

Centers for Disease Control and Prevention (CDC). US Department of Human Health and Services (DHHS). 2009:

- a) Swine Influenza A (H1N1) Infection in Two Children, Southern California, March/April 2009. *Morbidity and Mortality Weekly Report* (MMWR). 58 (April 24): 15: pp400–402.
- b) Outbreak of Swine-Origin Influenza A (H1N1) Virus Infection. *Morbidity and Mortality Weekly Report* (MMWR). April 30: 58 (Dispatch); 1-3.
- c) Serum Cross-Reactive Antibody Response to a Novel Influenza A (H1N1) Virus after Vaccination with Seasonal Influenza Vaccine. *Morbidity and Mortality Weekly Report* (MMWR). 58: 19: pp521–524.

Centers for Disease Control and Prevention (CDC), US Department of Human Health and Services (DHHS).

- a) NIOSH. Publications and Products. Mixed Exposures Research Agenda A Report by the NORA Mixed Exposures Team. DHHS (NIOSH) Publication Number 2005 106 http://www.cdc.gov/niosh/docs/2005-106/#exec (accessed December 2011).
- b) Vaccine Excipients and Media Summary. <a href="http://www.cdc.gov/vaccines/pubs/pinkbook/downloads/appendices/B/excipient-table-2.pdf">http://www.cdc.gov/vaccines/pubs/pinkbook/downloads/appendices/B/excipient-table-2.pdf</a> (accessed March 2012).
- c) Vaccines and Immunizations. Vaccines and Preventable Diseases (VPD).

- http://www.cdc.gov/vaccines/vpd-vac/default.htm (accessed July 2012).
- d) Vaccine Adverse Events Reporting System. VAERS. <a href="http://www.cdc.gov/vaccinesafety/Activities/vaers.html">http://www.cdc.gov/vaccinesafety/Activities/vaers.html</a> (accessed March 2013).
- e) Forrester FT. The Iceberg Concept of Infection. Public Health Image Library.

Chan M. 2010. WHO Director-General replies to the BMJ (letter). *British Medical Journal*. 340 (June): c3463. doi:10.1136/bmj.c3463.

Chee G, Molldrem V, Hsi N, Chankova S. 2008. Evaluation of the GAVI Phase 1 Performance (2000-2005). Bethesda MD. Abt Associates Inc.

Chen S, Yang Y, Yan X, Chen J, Yu H, Wang W. 2012. Influence of vitamin A status on the antiviral immunity of children with hand, foot and mouth disease. *Clinical Nutrition*. 31: 4: pp543-548.

Clifford GM, Smith JS, Plummer M, Munoz N, Franceschi S. 2003. Human Papillomavirus types in invasive cervical cancer worldwide: a meta-analysis. *British Journal of Cancer*. 88 (Jan): pp63-73.

Clifford GM, Gallus S, Herrero R, Muñoz N, Snijders PJF, Vaccarella S, Anh PTH, Ferreccio C, Hieu NT, Matos E, Molano M, Rajkumar R, Ronco G, de Sanjosé S, Shin HR, Sukvirach S, Thomas JO, Tunsakul S, Meijer CJLM, Franceschi S, The IARC HPV Prevalence Surveys Study Group et al. 2005. Worldwide distribution of human papillomavirus types in cytologically normal women in the International Agency for Research on Cancer HPV prevalence surveys: a pooled analysis. *The Lancet*. 366: 9490 (Sept): pp991-998.

Cochrane A. 1989. Archie Cochrane in his own words: Selections arranged from his 1972 introduction to "effectiveness and efficiency: Random reflections on the health services." *Controlled Clinical Trials.* 10: 4 (Dec): pp428–433.

Cogliano V, Baan R, Straif K, Grosse Y, Secretan B. 2005. Carcinogenicity of human papillomaviruses. *Lancet Oncology*. 6: 4 (April): p204.

Cohen D and Carter. 2010. WHO and the pandemic flu 'conspiracies'. *British Medical Journal*. 340 (June): c3257 doi: 10.1136/bmj.c257.

Colgrove J. 2006. State of Immunity: The Politics of Vaccination in Twentieth Century America. Berkeley: University of California Press.

Collignon P, Doshi P, Jefferson T. 2010. Adverse events following influenza vaccination in Australia – should we be surprised? *BMJ Rapid Responses*. 7 May.

Collins Solicitors. 2009. Corby Children Win Landmark Toxic Waste Case. *Press Release*. 29 July. <a href="http://www.collinslaw.co.uk/media.asp?id=229">http://www.collinslaw.co.uk/media.asp?id=229</a> (accessed Nov 2012).

Commonwealth of Australia (CoA). 1945–1986. Official Yearbook of the Commonwealth of Australia, No.37-72.

Commonwealth of Australia (CoA) 1999, National Environmental Health Strategy, Australian Government Publishing Service. Canberra. Cited in Stoneham et al 2004.

Confavreux C, Suissa S, Saddier P, Bourdès V, Vukusic S. 2001. Vaccinations and the Risk of Relapse in Multiple Sclerosis. *New England Journal of Medicine*. 344 (February): pp319-326.

Corderoy A. 2010. "Flu shot side effects 'worse than the disease". *Sydney Morning Herald Health*. 18<sup>th</sup> September.

Coulter HL. 1990. *Vaccination, Social Violence and Criminality: The Medical Assault on the American Brain*. Berkeley: North Atlantic Books: Washington DC: Center for Empirical Medicine.

CSL Ltd. 2007. Package Insert Afluria (Fluvax/Enzira). Influenza Virus Vaccine. <a href="http://novaccine.com/pdffiles/afluria\_flu\_influenza\_vaccine\_package\_insert.pdf">http://novaccine.com/pdffiles/afluria\_flu\_influenza\_vaccine\_package\_insert.pdf</a> (accessed March 2013).

CSL Limited. 2008. Annual Report 2007-8; Year in Review. http://www.csl.com.au/docs/565/929/CSL\_AR\_2008.pdf#search=year in review 2008-9 (accessed March 2014).

CSL Limited, Research and Development, Recent Highlights, Gardasil http://www.csl.com.au/research-development/recent-highlights.htm (accessed Nov 2011)

Cumpston JHL. 1989. Ed. Lewis MJ. *Health and Disease in Australia: A History by JHL Cumpston*. Canberra: Australian Government Publishing Service.

Curry L. 2002. The Human Body on Trial. California: ABC-CLIO Inc.

Dayan GH, Quinlisk MP, Parker AA, Barskey AE, Harris ML, Schwartz JM, Hunt K, Finley CG, Dennis BS, Leschinsky DP, O'Keefe AL, Clayton J, Kightlinger LK, Dietle EG, Berg J, Kenyon CL, Goldstein ST, Stokley SK, Redd SB, Rota PA, Rota J, Bi D, Roush SW, Bridges CB, Santibanez TA, Parashar U, Bellini WJ, Seward JF. 2008. Recent resurgence of mumps in the United States. *New England Journal of Medicine*. 358 (April): pp1580-1589. Dayton L. 2011. Ian Frazer's HPV vaccine may prevent cervical cancer. *The Australian*. 19<sup>th</sup> June

Dean T. 2009. CSL Speaks out on Vaccination 'Myths'. Australian Life Scientist. October 1st.

De Serres G, Markowski F, Toth E, Landry M, auger D, Mercier M, Belanger P, Turmel B, Arruda H, Boulianne N, Ward BJ, Skowronski DM. 2012. The Largest measles epidemic in North America in a decade – Quebec. Canada. 2011. Contribution of susceptibility, serendipity and super-spreading events on elimination. *The Journal of Infectious Diseases*. 207: 6: pp990-998.

Doherty P. 2010. No let-up in crusade against killer flu: influenza A. *The Australian*. 2<sup>nd</sup> October.

Downing D. 2011. The Health Hazards of Disease Prevention. *Orthomolecular News Service Press Release April 2011*. Cited in the British Society For Ecological Medicine (BSEM). Health Hazards. <a href="http://www.ecomed.org.uk/publications/the-health-hazards-of-disease-prevention/orthomolecular-news-service-press-release">http://www.ecomed.org.uk/publications/the-health-hazards-of-disease-prevention/orthomolecular-news-service-press-release</a> (accessed March 2013).

Downing NS, Aminawung JA, Shah ND, Krumholz HM, Ross JS. 2014. Clinical Trial Evidence Supporting FDA Approval of Novel Therapeutic Agents, 2005-2012. *Journal of the American Medical Association*. 311: 4: pp368-377.

Doyal L and Doyal L. 1984. Western Scientific Medicine: a philosophical and political prognosis. In Birke L and Silvertown J. (eds.) *More than the Parts: Biology and Politics*. London: Pluto Press Ltd. Chapter 5.

Dubos R, Pines M. and the editors of Life. 1965. Health and Disease. New York: Time Inc.

Duclos P. 2010. National Immunisation Technical Advisory Groups (NTIAGs): Guidance for their establishment and strengthening. *Vaccine*. 28 (April) Suppl 1: A18-A25.

Dwyer J and Eagar K. 2008. Options for reform of the Commonwealth and State governance responsibilities for the Australian Health System. A paper commissioned by the National Health and Hospitals Reform Commission.

Dyer C. 2009. Judge Rules that Birth Defects Could Have Been Caused by Toxic Waste from Steel Works. *British Medical Journal*. 339 (August): p3162.

Eduljee G H. 2000. Trends in Risk Assessment and Risk Management. *The Science of the Total Environment*. 249: 1-3(April): pp13-23.

Eklund C, Forslund O, Wallin KL, Zhou T, Dillner J. 2012. WHO Human Papillomavirus Laboratory Network: The 2010 global proficiency study of human papillomavirus genotyping in vaccinology. *Journal of Clinical Microbiology*. 50: pp2289-98.

Eldred BE, Dean AJ, McGuire TM, Nash AL. 2006. Vaccine Components and constituents: responding to consumer concerns. *Medical Journal of Australia*. February. 184: 4: pp170-175.

Engdahl FW. 2015. Shocking Report from Medical Insiders. NSNBC International. 19 June.

Epstein H. 2011. Flu Warning: Beware the Drug Companies. *The New York Review of Books*. <a href="http://www.nybooks.com/articles/archives/2011/may/12/flu-warning-beware-drug-companies/">http://www.nybooks.com/articles/archives/2011/may/12/flu-warning-beware-drug-companies/</a> (accessed October 11)

Epstein SS. 1979. The Politics of Cancer. New York: Anchor/Doubleday

European Commission (EC). 2008. Anon. Strengthening pharmacovigilance to reduce adverse effects of medicines. Brussels:

http://ec.europa.eu/health/files/pharmacos/pharmpack\_12\_2008/memo\_pharmacovigiliance\_december\_2008\_en.pdf (accessed August 2012).

European Medicines Agency (EMA). 2009. EMEA: Pandemic influenza A (H1N1)v vaccines authorized via the core dossier procedure. Explanatory note on scientific considerations regarding the licensing of pandemic A(H1N1)v vaccines. London. 24 September .

Erikson N. 2014. Breaking News from Japan: International Symposium on the Adverse Reactions Experienced by those Vaccinated with Human Papillomavirus Vaccines. S.A.N.E. VAX Inc. February 25<sup>th</sup> <a href="www.sanevax.org">www.sanevax.org</a> (accessed March 2014).

Feery B. 1981. Impact of Immunisation on Disease Patterns in Australia. *Medical Journal of Australia*. August. 2: 4: pp172-176.

Feery BJ. 1982. Incidence and type of reactions to triple antigen (DTP) and DT (CDT) vaccines. *Medical Journal of Australia*. 2: pp511-515.

Fidler DP and Gostin LO. 2006. The New International Health Regulations: An Historical Development in Law and Public Health. *Journal of Law, Medicine and Ethics*. 34: pp85-94.

Finucane ML. 2004. The Psychology of Risk Judgments and Decisions. In Cromar N, Cameron S and Fallowfield H. (eds.) *Environmental Health in Australia and New Zealand*. Melbourne: Oxford University Press. pp142-155.

Fitzgerald PD. 2001. The ethics of doctors and big business. *Medical Journal of Australia*. 175: 2: pp73-75.

Flynn P. 2010. The handling of the H1N1 pandemic: more transparency needed. Parliamentary Assembly Council of Europe. Social Health and Family Affairs committee. United Kingdom.

Food and Drug Administration (FDA). 2006. FDA Licenses New Vaccine for Prevention of Cervical Cancer and Other Diseases in Females Caused by Human Papillomavirus: Rapid Approval Marks Major Advancements in Public Health. US Department of Health and Human Services. USA. News Release. June 8.

Food and Drug Administration. 2008. CDC Purchased 10 Million Doses of Gardasil®. http://www.fdanews.com/newsletter/article?articleId=103084&issueId=11210 (accessed February 2013).

Food and Drug Administration (FDA). Vaccines, Blood and Biologics. Thimerosal in Vaccines. US Department of Health and Human Services. <a href="http://www.fda.gov/BiologicsBloodVaccines/SafetyAvailability/VaccineSafety/UCM096228">http://www.fda.gov/BiologicsBloodVaccines/SafetyAvailability/VaccineSafety/UCM096228</a> (accessed March 2013)

Food and Drug Administration (FDA). 1982. Mercury-containing products for topical antimicrobial over-the counter human use; Establishment of a Monograph, proposed rules, Federal Register 47/436-01. January 5. Cited in Kirby 2005. *Evidence of Harm. Mercury in Vaccines and the Autism Epidemic: A Medical Controversy*. New York: St. Martin's Press. p83.

Fourrier A, Touze E, Alperovitch A, Begaud B. 1999. Association Between Hepatitis B Vaccine and Multiple Sclerosis: A Case-control Study. *Pharmacoepidemiology and Drug Safety.* 8: Suppl: S140-S141.

Franco E. 1995. Cancer Causes Revisited: Human Papillomavirus and Cervical Neoplasia, *Journal of the National Cancer Institute*. 87: 11(June): pp779-80.

Fraser Health. 2010. VAXIGRIP Inactivated Influenza Vaccine Trivalent Types A and B (Split Virion). Sanofi-pasteur Ltd.

Freidson E. 1970. *Profession of Medicine: A Study of the Sociology of Applied Knowledge*. New York: Harper and Row.

Freidson E. 1986. *Professional Powers: a study of the institutionalization of formal knowledge*. Chicago and London: University of Chicago Press.

Frickel S, Gibbon S, Howard J, Kempner J, Ottinger G, Hess D. 2010. Undone Science: Charting Social Movement and Civil Society Challenges to Research Agenda Setting. *Science, Technology and Human Values.* 35: 4(July): pp444-473.

Frickel S and Moore K (ed). 2006. *The New Political Sociology of Science: institutions, networks and power*. Madison, WI: The University of Wisconsin Press.

Friis RH and Sellers TA. 2004. *Epidemiology for Public Health Practice* (3<sup>rd</sup> Ed). Massachusetts: Jones and Bartlett Publishers.

Future II Study Group. 2007. Quadrivalent Vaccine against Human Papillomavirus to Prevent High-Grade Cervical Lesions. *The New England Journal of Medicine*. 356: 19: pp1915–27.

Gallagher CM and Goodman CS. 2010. Hepatitis B vaccination of male neonates and autism diagnosis, NHIS 1997-2002. *Journal of Toxicology and Environmental Health*. 73: 24: pp1665-77.

Garrett L. 2012a. Drug Safety Crisis, *Australian Broadcasting Commission (ABC) Saturday Extra*, Council on Foreign Relations. May 19. <a href="http://mpegmedia.abc.net.au/rn/podcast/2012/05/sea\_20120519\_0745.mp3">http://mpegmedia.abc.net.au/rn/podcast/2012/05/sea\_20120519\_0745.mp3</a> (accessed November 2012).

Garrett L. 2012b. Innovation Memorandum No 21: Ensuring the Safety of the World's Drug, Vaccine and Medicines Supply. Council on Foreign Relations. Press Release. May. <a href="http://www.cfr.org/global-health/ensuring-safety-integrity-worlds-drug-vaccine-medicines-supply/p28256">http://www.cfr.org/global-health/ensuring-safety-integrity-worlds-drug-vaccine-medicines-supply/p28256</a> (accessed November 2012).

Gessner BD, Duclos P, DeRoeck D, Nelson EAS. 2010. Informing decision makers: Experience and process of 15 National Immunization Technical Advisory Groups. *Vaccine*. 28 (April): Suppl 1: ppA1-A5.

Geier D, Geier MR, Geier D, Geier MR. 2005. A case-control study of serious autoimmune adverse events following hepatitis B immunization. *Autoimmunity*. 38: 4: pp295-301.

Gerathy S. 2013. Childcare centres face fines for enrolling unvaccinated children. ABC News. May.

Gilbert SG. 2004. *A Small Dose of Toxicology: the health effects of common chemicals.* Florida: CRC Press.

Gillespie JA. 1991. *The Price of Health: Australian Governments and Medical Politics 1910–1960.* Cambridge: Cambridge University Press.

Gitsch G, Kainz Ch, Reinthaller A, Kopp W, Tatra G, Breitenecker G. 1991. Cervical neoplasia and human papilloma virus infection in prostitutes. *Genitourin Medicine*. December. 67: 6: pp478-480.

GlaxoSmithKline Biologicals. 2005. Prescribing Information Engerix-B (Hepatitis B Recombinant Vaccine). <a href="http://www.novaccine.com/pdffiles/engerix-b\_prescribing\_information\_leaflet.pdf">http://www.novaccine.com/pdffiles/engerix-b\_prescribing\_information\_leaflet.pdf</a> (accessed June 2012). Global Alliance for Vaccines and Immunisation (GAVI), History of GAVI (HoG) <a href="http://www.gavialliance.org/about/mission/history/">http://www.gavialliance.org/about/mission/history/</a> (accessed July 2014)

Goldacre B. 2012. *Bad Pharma: How Drug Companies Mislead Doctors and Harm Patients*. London: Fourth Estate.

Goldsmid J. 1988. *The Deadly Legacy: Australian History and Transmissible Disease*. Sydney: New South Wales University Press.

Government of Western Australia (GWA), Department of Health:

- a) Health Risk Assessment in Western Australia. 2006.
- b) Communicable Diseases Control Directorate. 2008. June.
- c) Communicable Diseases Control Directorate. Influenza Fact Sheet (IFS). 2009.
- d) Communicable Diseases Control Directorate. 2009.
- e) Public Health Act 2008 (draft).

Granoff DM and Rappuoli R. 1997. Are serological responses to acellular pertussis antigens sufficient criteria to ensure that new combination vaccines are effective for prevention of disease? *Developments in Biological Standardisation*. 89: pp379-89.

Greville, RW. 1966. Recent and Future Development in Immunising Vaccines. *The Medical Journal of Australia*. May: p908.

Grigor W. 1965. Research in Immunology. The Medical Journal of Australia. 1: p65 and p83.

Gronen JA. Harris GD. Harper DM. 2014. Reduction in HPV Prevalence – No Evidence to support HPV Vaccination reduces HPV Prevalence, *Journal of Infectious Diseases*, Correspondence, 209: 8: pp1302-1304

Gross M. 2007. The Unknown in Process: Dynamic connections of Ignorance, Non-Knowledge, and Related Concepts. *Current Sociology*. 55: 5: pp742-59.

Guido F, Duclos P, Margolis H, Lavanchy D, Siegrist C, Meheus A, Lambert PH, Emiroglu N, Badur S, Van Damme P. 2005. Vaccine Safety Controversies and the Future of Vaccination Programs. *The Pediatric Infectious Disease Journal*. 24 (November):11: pp953-961.

Gustafsson L. Pontén J, Bergström R, Adami HO. 1997. International incidence rates of invasive cervical cancer before cytological screening. *International Journal of Cancer*. 71: pp159–165.

Habakus LK. 2011. A Human Rights Assessment. In Habakus and Holland M. (eds). *Vaccine Epidemic: how corporate greed, biased science and coercive government threaten our human rights, our health and our children*. New York: Skyhorse. Chapter 3.

Habakus LK and Holland M. (eds) 2011. *Vaccine Epidemic: how corporate greed, biased science and coercive government threaten our human rights, our health and our children.* New York: Skyhorse.

Hansen J. 2012. "Parents fight it out in tussle over jabs." *Sunday Telegraph*. June 24<sup>th</sup>.

Hartley J. 2013. Anti-vax parents to be known as 'refusers'. *Australian Doctor.com*. June 7. <a href="http://www.australiandoctor.com.au/news/latest-news/anti-vax-parents-to-be-known-as-refusers">http://www.australiandoctor.com.au/news/latest-news/anti-vax-parents-to-be-known-as-refusers</a> (accessed October 2013).

Harris P. 2006. "Capitol Hill's dirty secrets laid bare." *Guardian Weekly*. January 13-19. p6.

Haug C. 2009. The risks and Benefits of HPV Vaccination. *Journal of the American Medical Association*. 302: 7: pp795-796.

Haverkos H. 2005. Multifactorial Etiology of Cervical Cancer: A Hypothesis. *Medscape General Medicine*. 7: 4: p57.

Hawe P.1994. Measles Control: a best practice challenge in Public Health. *Australian Journal of Public Health*. 18: 3(Sept): pp241-243.

Hawkes D, Lea C, Berryman M. 2013. Answering human papillomavirus vaccines concerns: a matter of science and time. *Infectious Agents and Cancer*. 8: 22 (June): pp1-8.

Hays JN. 2000. *The Burden of Disease: Epidemics and Human Response in Western History*. New Brunswick, NJ: Rutgers University Press.

Heidary H and Cohen D. 2005. Hypersensitivity Reactions to Vaccine Components. *Dermatitis*. 16: 3: pp115-120.

Heikkinen T, Booy R, Campins M, Finn A, Olcen P, Peltola H, Rodrigo C, Schmitt H, Schumacher F, Teo S, Weil-Olivier C. 2006. Should healthy children be vaccinated against influenza? *European Journal of Pediatrics*. 165: pp223-228. DOI 10.1007/s00431-005-0040-9.

Henderson BE. 1989. Editorial. Establishment of an Association Between a Virus and a Human Cancer. *Journal of the National Cancer Institute*. 81: 5: pp320-321.

Herskovits B. 2007. Brand of the Year. *Pharmaceutical Executive*. 27: 2: pp58-65.

Hess DJ. 2007. *Alternative Pathways in Science and Industry: activism, innovation and the Environment in an Era of Globalisation*. Cambridge, MA: Massachusetts Institute of Technology Press.

Hess D. 2009. The Potentials and Limitations of civil Society Research: Getting Undone Science Done. *Sociological Inquiry*. 79: 3(Aug): pp306–327.

Hetzel BS. 1976. Health and Australian Society. Victoria. Penguin Books Australia Ltd.

Hicks N. 2004. The Politics and Practice of Environmental Reform. In Cromar N, Cameron S, Fallowfield H. (eds). *Environmental Health in Australia and New Zealand*. Melbourne: Oxford University Press. pp156-166.

Hobbins P. 2011. Immunisation is as popular as a death adder: The Bundaberg tragedy and the politics of medical science in interwar Australia. *Social History of Medicine*. 24: 2: pp.426–444.

Holland M. 2011. Vaccination choice is a fundamental human right. In Habakus LK and Holland M. (eds). *Vaccine Epidemic: how corporate greed, biased science and coercive government threaten our human rights, our health and our children*. New York: Skyhorse. Chapter 1.

Homeland Security America. 1957 Asian Flu Pandemic and 1976 Swine Flu Pandemic Scare. www.GlobalSecurity.org (accessed October 2010).

Howett MK, Kuhl JP. 2005. Microbicides for Prevention of Transmission of Sexually Transmitted Diseases. *Current Pharmaceutical Design.* 11: pp3731-3746.

Hui Ming Yang, Meng M, Chaomin W. 2005. Vitamin A for treating measles in children (Review). *Cochrane Database of Systematic Reviews*. CD001479. October 19.

Iannacone M, Moseman EA, Tonti E, Bosurgi L, Junt T, Henrickson SE, Whelan SP, Guidotti LG, Luca G, von Adrian UH, von Adrian G. 2010. Subcapsular sinus macrophages prevent CNS invasion on peripheral infection with a neurotropic virus. *Nature*. 465 (June): pp1079-1083.

Iannuzzi A. 2012. Immunisation Confessions. *Medical Journal of Australia Insight*. Online Journal. <a href="http://www.mjainsight.com.au/view?post=aniello-iannuzzi-immunisation-confessions&post\_id=8919&cat=comment">http://www.mjainsight.com.au/view?post=aniello-iannuzzi-immunisation-confessions&post\_id=8919&cat=comment</a> (accessed April 2012).

Illich I. 1976. Medical Nemesis: The Expropriation of Health. London: Calder and Boyars L.

Immunisation Action Coalition. IAC Mission Statement. <a href="www.immunize.org">www.immunize.org</a> (accessed May 2013).

Institute of Medicine (IOM). 1993. *The Children's Vaccine Initiative (CVI): Achieving the Vision*. Washington DC: The National Academy of Sciences.

Institute of Medicine (IOM). 2002. *The future of Public Health in the 21<sup>st</sup> Century*. Washington DC: National Academies Press.

Influenza Specialist Group (ISG). <a href="www.influenzaspecialistgroup.org.au">www.influenzaspecialistgroup.org.au</a> (accessed July 2011).

International Agency for Research on Cancer (WHO) (IARC). 1995. IARC Monographs on the Evaluation of Carcinogenic Risks to Humans. Volume 64. Human Papillomaviruses.

Ioannidis JPA. 2005. Why Most Published Research Findings Are False. *PloS Med:* 2: p2124.

James W. 1988. *Immunization: the reality behind the myth.* Westport, CT: Bergin & Garvey.

Jefferson T, Rivetti D, Di Pietrantonj C, Rivetti A, Demicheli V. 2007. Vaccines for preventing influenza in healthy adults. *Cochrane Database of Systematic Reviews*. Issue 2. Art No: CD001269.

Jefferson T. 2009. Interview with Tom Jefferson. A Whole Industry is Waiting for a Pandemic. *Der Spiegel*. 21 July 2009 as cited in Flynn. 2010. The handling of the H1N1 pandemic: more transparency needed. Parliamentary Assembly Council of Europe. Social Health and Family Affairs committee. United Kingdom. p8.

John Hopkins University and Bloomberg School of Public Health. 2005. Disease Warriors. Television Program. Part 1–4. Holt Production Films. Boston: WGBH/NOVA Science Unit and Vulcan Productions Inc. Arlington.VA.

Katz J, Hancock K, Veguilla V, Zhong W, Lu XH, Sun H, Butler E, Dong L, Liu F, Li ZN, DeVos J, Gargiullo P, Cox N. 2009. Serum cross-reactive antibody response to a novel influenza A (H1N1) virus after vaccination with seasonal influenza vaccine. *Morbidity and Mortality Weekly Report* (MMWR). 58: 19: pp521–524.

Kirby D. 2005. Evidence of Harm: Mercury in Vaccines and the Autism Epidemic: A Medical Controversy. New York: St. Martin's Press.

Klein NP, Bartlett J, Rowhani-Rahbar A, Fireman B, Baxter R. 2012. Waning Protection after Fifth Dose of Acellular Pertussis Vaccine in Children. *New England Journal of Medicine*. 367(Sept): pp1012-1019.

http://www.nejm.org/doi/full/10.1056/NEJMoa1200850

Kleinman DL, Kinchy AJ, & Handelson J. Eds. 2005. *Controversies in Science and Technology Volume 1: From Maize to Menopause*. Madison, WI: The University of Wisconsin Press

Klish WJ, Baker SS. 1996. Aluminium Toxicity in Infants and Children. *Pediatrics*. 97: 3: p413.

Klug SJ, Molijn A, Schopp B, Holz B, Iftner A, Quint WJ et al. 2008. Comparison of the performance of different HPV genotyping methods for detecting genital HPV types. *Journal of Medical Virology*. 80: 7(July): pp1264-74.

Kmietowicz Z. 2010. WHO admits to 'inconsistencies' in its policy on conflicts of interest. *British Medical Journal*. 240 (15 June): c3167 doi: 10.1136/bmj.c3167.

Kothalawala H, Toussaint M, Gruys E. 2006. An Overview of Swine Influenza. Vet Q 28 (2): pp46–53. PMID 16841566.

Krimsky S. 2003. *Science in the Private Interest: has the lure of profits corrupted biomedical research?* Lanham MD: Rowman and Littlefield.

Lancaster HO. 1956:

- a) Infant Mortality in Australia. *The Medical Journal of Australia*. 2: pp100-108.
- b) The Mortality of Childhood in Australia: Part 1 Early Childhood. *Medical Journal of Australia*. 2: pp889-894.

Langley A. 2004. Risk Assessment. In Cromar N, Cameron S, Fallowfield H. (eds). *Environmental Health in Australia and New Zealand*. Melbourne: Oxford University Press. pp92-110.

Langmuir AD, Henderson RE, Serfling RE, Sherman IL. 1962. The importance of measles as a health problem. *The American Journal of Public Health and the Nations Health*. 52: 2(Feb): Suppl:1-4.

Laurence E and McLaren N. 2014. Anti-vaccination storm brewing at UOW. ABC Health Report. February 1.

Lee BR, Feaver SL, Miller CA, Hedberg CW, Ehresmann KR. 2004. An elementary school outbreak of varicella attributed to vaccine failure: policy implications. *The Journal of Infectious Diseases*. 190: 3: pp477-483.

Lopez AS, Guris D, Zimmerman L, Gladden L, Moore T, Haselow DT, Loparev VN, Schmid DS, Jumaan AO, Snow SL. 2006. One dose of varicella vaccine does not prevent school outbreaks: is it time for a second dose? *Pediatrics*. 117: e1070-e1077.

Magill TP. 1955. The Immunologist and the Evil Spirits. *The Journal of Immunology*. 74: 1: pp1–8.

Marshall E. 1998. A Shadow Falls on the Hepatitis B Vaccination Effort. *Science*. 281: 5377: pp630-1.

Martin B. 2015. On the Suppression of Vaccination Dissent. *Science and Engineering Ethics*. 21: 1: pp143-157.

Martin E. (ed). 2002. *Oxford Concise Colour Medical Dictionary* (Third Ed). Oxford: Oxford University Press.

McCredie MRE, Sharples CP, Paul C, Baranyai J, Medley G, Jones RW, Skegg DCG. 2008. Natural history of cervical neoplasia and risk of invasive cancer in women with cervical intraepithelial neoplasia 3: a retrospective cohort study. *Lancet Oncology*. 9: 5: pp425-434

McIntyre P. 2012. Does whooping cough vaccine for parents protect newborns and who should pay for it? *The Conversation*. May. <a href="https://theconversation.com/does-whooping-cough-vaccine-for-parents-protect-newborns-and-who-should-pay-for-it-6980">https://theconversation.com/does-whooping-cough-vaccine-for-parents-protect-newborns-and-who-should-pay-for-it-6980</a> (accessed June 2012).

McKee M and Britton A. 1997. Conducting a literature review on the effectiveness of health care interventions. *Health Policy and Planning*. 12: 3: pp262-267.

McKeown T. 1979. *The role of Medicine: Dream, Mirage or Nemesis?* Oxford: Basil Blackwell.

McNeill D, Andresen S and Sandberg K. 2013. The global politics of health: actors and initiatives. In Roalkvam S, McNeill D and Blume S (eds). *Protecting the World's Children: Immunisation Policies and Practice*. Oxford. Oxford University Press. pp59-87.

Medical Board of Australia (MBA). 2010. Good Medical Practice: A Code of Conduct for Doctors in Australia. <a href="http://www.medicalboard.gov.au/Codes-Guidelines-Policies.aspx">http://www.medicalboard.gov.au/Codes-Guidelines-Policies.aspx</a> (accessed October 2013).

Medical Products Agency (MPA). 2007. Public Assessment Report. Scientific Discussion. Afluria, suspension for injection, Influenza vaccine (split virion, inactivated). Mutual Recognition Procedure. SE/H/o485/01/E01. Sweden. June 28. <a href="http://www.lakemedelsverket.se/SPC\_PIL/Pdf/par/Afluria,%20suspension%20for%20injection.pdf">http://www.lakemedelsverket.se/SPC\_PIL/Pdf/par/Afluria,%20suspension%20for%20injection.pdf</a> (accessed 2010).

Mercae A. 2003. From Immunology to Social Policy: Epistemology and Ethics in the Creation and Administration of Paediatric Vaccines. PhD Thesis. University of Tasmania.

Merck & Co. Gardasil Package Insert. Reporting Adverse Events: Page 1. (updated March 2013). http://www.merck.com/product/usa/pi\_circulars/g/gardasil/gardasil\_pi.pdf

Merck &Co. 2011. Strategic National Stockpile of Gardasil®. 10-K SEC. <a href="http://phx.corporateir.net/External.File?item=UGFyZW50SUQ9ODM1MDV8Q2hpbGRJRD">http://phx.corporateir.net/External.File?item=UGFyZW50SUQ9ODM1MDV8Q2hpbGRJRD</a> 0tMXxUeXBlPTM=&t=1. (accessed February 2013).

Merck, Sharp and Dohme Corporation (MSD). Subsidiary of Merck and Co Inc Partnering with Australian Researchers and Innovators.

http://www.msdaustralia.com.au/content/corporate/index.html (accessed November 2011).

Merck, Sharp and Dohme Corporation. Research Partnerships. A Track Record of Successful Research Partnerships.

http://www.msdaustralia.com.au/content/corporate/research/Partnerships.html (accessed November 2011).

Michaels D. 2008. *Doubt is Their Product: How Industry's Assault on Science Threatens Your Health*. New York: Oxford University Press.

Mihalovic D. 2012. Recent Evidence shows Vaccinated Kids Account for 90% of cases of Whooping Cough. Prevent Disease <a href="http://preventdisease.com/news/12/113012\_Vaccinated-Kids-Account-For-90-Percent-of-Cases-of-Whooping-Cough.shtml">http://preventdisease.com/news/12/113012\_Vaccinated-Kids-Account-For-90-Percent-of-Cases-of-Whooping-Cough.shtml</a> (accessed September 2013).

Mokhiber R. 1989. Corporate crime and Violence: Big Business Power and the Abuse of the Public Trust. San Francisco: Sierra Club Books.

Molina V and Schoenfeld Y. 2005. Infection, vaccines and other environmental triggers of autoimmunity. *Autoimmunity*. 38(May): pp235-245.

Mori S, Nakao S, Kukimoto I, Kusumoto-Matsuo R, Kondo K, Kanda T. 2011. Biased amplification of human papillomavirus 16 and 18 DNA in specimens containing multiple human papillomavirus types by PCR with consensus primers. *Cancer Science*. 102: 6(June): pp1223-27.

Moseman EA, Iannacone M, Bosurgi L, Tonti E, Chevrier N, Tumanov A, Fu YX, Hacohen N, von Andrian UH. 2012. B cell maintenance of subcapsular sinus macrophages protects against a fatal viral infection independent of adaptive immunity. *Immunity*. 36: 3: pp415-426.

Moser KA, Leon DA and Gwatkin DR. 2005. How does progress towards the child mortality millennium development goal affect inequalities between the poorest and least poor? Analysis of Demographic and Health Survey data. *British Medical Journal*. 331 (November): pp1180-1182.

Munoz N, Castellagnue X, Berrington de Gonzales A, Gissmann L. 2006. HPV Vaccines and Screening in the Prevention of Cervical Cancer. Ch. 1: HPV in the Etiology of Human Cancer. *Vaccine*. 24(Aug): Suppl 3: ppS1-S10.

Munoz N, Bosch F, Jensen O. (eds). 1989. *Human Papillomavirus and Cervical Cancer*. International Agency for Research on Cancer (WHO). Lyon: Scientific Publications No.94.

Muraskin W. 1998. The Politics of International Health: The Children's vaccine Initiative and the Struggle to Develop Vaccines for the Third World. Albany NY: State University of New York Press.

Muraskin W. 2004. The Global Alliance of Vaccines and Immunisation: is it a new model for effective public-private cooperation in international public health? *American Journal of Public Health.* 94 (Nov): 11: pp1922-5.

Murdoch University. The Institute for Immunology and Infectious Diseases. Genesiis Program. Western Australia.

Naidoo J and Wills J. 2000. *Health Promotion: Foundations for Practice* (Second Ed). London: Harcourt.

National Australia Day Council (NADC). Australian of the Year Awards Honour Roll. Australian of the Year 2006. <a href="http://www.australianoftheyear.org.au/honour-roll/?view=fullView&recipientID=109">http://www.australianoftheyear.org.au/honour-roll/?view=fullView&recipientID=109</a> (accessed June 2013).

NCIRS. 2009. Human Papillomavirus (HPV) vaccines for Australians. Australian Government. <a href="http://www.ncirs.edu.au/immunisation/fact-sheets/hpv-human-papillomavirus-fact-sheet.pdf">http://www.ncirs.edu.au/immunisation/fact-sheets/hpv-human-papillomavirus-fact-sheet.pdf</a> (updated March 2013).

Newall KW. 1988. Selective Primary Health Care: The Counter Revolution. *Social Science and Medicine*. 26: 9: pp903-906.

NHMRC. 1954–1986. Report of the Session No.38 – 101. Canberra: Commonwealth Government of Australia.

NHMRC. 1991. Commonwealth of Australia. *Immunisation Procedures* (fourth edition). Canberra: Australian Government Publishing Service.

NHMRC. 2003. *Australian Immunisation Handbook* (eighth edition). Commonwealth of Australia.

NHMRC. 2005. Screening to prevent cervical cancer: guidelines for the management of asymptomatic women with screen detected abnormalities. National Screening Program. Australian Government.

National Health Service (NHS) executive. 1992. Local Voices: the views of local people in purchasing for health. London: Green Print. In Naidoo and Wills. 2000. Chapter 9 p192.

Needleman H. 2000. Needleman Explores the Link Between Chemicals and Crime. The Heinz Awards. www.heinzawards.net/achievement (accessed March 2006).

Nkowane BM, Bart SW, Orenstein WA, Baltier M. 1987. Measles outbreak in a vaccinated school population: epidemiology, chains of transmission and the role of vaccine failures. *American Journal of Public Health*. 77: 4(April): pp434-438.

Nolan T. 2010. The Australian model of immunisation advice and vaccine funding. *Vaccine*. 28 (April) Suppl 1: A76-A83.

Nolan T, McVernon J, Skeljo M, Richmond P, Wadia U, Lambert S, Nissen M, Marshall H, Booy R, Heron L, Hartel G, Lai M, Basser R, Gittleson C, Greenberg M. 2010. Immunogenicity of a Monovalent 2009 Influenza A (H1N1) Vaccine in Infants and Children: A Randomised Trial. Journal of the American Medical Association. 303: 1(Jan): pp37-46: Supplementary online content. doi:10.1001/jama.2009.1911.

NSW Public Health Network. 2009. Progression and impact of the first winter wave of the 2009 pandemic H1N1 influenza in New South Wales, Australia. *Surveillance and outbreak report*, www.eurosurveillance.org/ViewArticle.aspx?ArticleID=19365.

Walsh N. 2014. Whooping cough cases rise among infant, raise concerns vaccine may be losing its effectiveness. <a href="http://www.abc.net.au/news/2014-11-27/concerns-rise-over-effectiveness-of-whooping-cough-vaccine/5923592">http://www.abc.net.au/news/2014-11-27/concerns-rise-over-effectiveness-of-whooping-cough-vaccine/5923592</a> (accessed November 2014).

Norris SL, Holmer HK, Ogden LA, Burda BU. 2011. Conflict of interest in clinical practice guideline development: a systematic review. *PLoS One*. 6(Oct): e25153.

NSW Department of Health. 2005. Policy Directive. Occupational Screening and Vaccination Against Infectious Diseases. PD2005 338.

NSW Department of Health. 2011. Policy Directive: Occupational Assessment, Screening and Vaccination against Specified Infectious Diseases. <a href="http://www.health.nsw.gov.au/policies/pd/2011/pdf/PD2011\_005.pdf">http://www.health.nsw.gov.au/policies/pd/2011/pdf/PD2011\_005.pdf</a> (accessed March 2012).

Obomsawin R. 1998. *Universal Immunization: Medical Miracle or Masterful Mirage* (Third Edition). Burnaby, B.C Canada: Health Action Network Society Publishers.

Ochsenbein AF, Pinschewer DD, Sierro S, Horvath E, Hengartner H, Zinkernagel RM. 2000. Protective long-term antibody memory bt antigen-driven and T help-dependent differentiation of long-lived memory B cells to short-lived plasma cells independent of secondary lymphoid organs. *Proceedings of the National Academy of Sciences USA*. 97: 24(Nov): pp13263-13268.

O'Dowd A. 2010. Council of Europe condemns 'unjustified scare' over swine flu. *British Medical Journal*. 340 (June): c3033: doi:10.1136/bmj.c3033

Olmsted D and Blaxill M. 2011. Counting Offit's Millions: More on How Merck's Rotary Vaccine made Paul Offit Wealthy. Age of Autism <a href="http://www.ageofautism.com/">http://www.ageofautism.com/</a> (Accessed December 2012).

O'Connor K. 1989. A History of 75 years of baby health services in NSW. NSW Department of Health.

Offit PA and Bell LM. 2003. *Vaccines: What You Should Know.* Third edition. Hoboken, NJ: John Wiley.

O'Neill Graeme. 2006. University of Queensland (UQ) team to test therapeutic HPV vaccine. *Australian Life Scientist*. February.

http://www.lifescientist.com.au/article/150021/uq\_team\_test\_therapeutic\_hpv\_vaccine/(accessed March 2012)

Palmer G and Short S. 1994. *Healthcare and Public Policy: An Australian Perspective*. Melbourne: Macmillan Education. pp5-20.

Palmer G and Short S. 2010. *Health Care and Public Policy: An Australian Analysis* (Fourth Ed). Melbourne: Palgrave Macmillan. Chs. 2 and 3.

Papania M, Baughman AL, Lee S, Cheek JE, AtkinsonW, Redd SC, Spitalny K, Finelli L, Markowitz L. 1999. Increased susceptibility to measles in infants in the United States. *Pediatrics*. 104: e59.

Parkes M and Weinstein P. 2004. An Ecosystems Approach. In Cromar N, Cameron S and Fallowfield H. (eds). *Environmental Health in Australia and New Zealand*. Melbourne: Oxford University Press. pp45-65.

Parkin DM, Bray F, Ferlay J, Pisani P. 2005. Global Cancer Statistics 2002. *CA: A Cancer Journal for Clinicians*. 55: 2: pp74-108.

Parliamentary Assembly Council of Europe (PACE). 2007. Recommendation 1787. The Precautionary principle and responsible management. <a href="www.assembly.coe.int/">www.assembly.coe.int/</a> (accessed November 2010).

Pearce N and Woodward A. 2004. Environmental Epidemiology. In Cromar N, Cameron S and Fallowfield H. (eds). *Environmental Health in Australia and New Zealand*. Melbourne: Oxford University Press. pp3-19.

Pelling M. 2002. The Meaning of Contagion: reproduction, medicine and metaphor. In Bashford A and Hooker C. (eds). *Contagion: epidemics, history and culture from smallpox to anthrax*. South Australia: Hyde Park Press.

Peterson M. 2008. Our Daily Meds. New York: Sarah Crichton Books

Pfister H. (ed). 1990. Papillomaviruses and Human Cancer. USA: CRC Press.

Pifferi G and Restani P. 2003. The Safety of Pharmaceutical Excipients. *Il Farmaco*. 58: pp541-550.

Public Health Association of Australia (PHAA). 2006. PHAA Newsletter. 23: 8 September. (accessed October 2006).

Public Health Agency of Canada (PHAC). 2007. Thimerosal Updated Statement. Communicable Disease Report. 33(July): ACS-6. Canada. <a href="http://www.phac-aspc.gc.ca/publicat/ccdr-rmtc/07vol33/acs-06/">http://www.phac-aspc.gc.ca/publicat/ccdr-rmtc/07vol33/acs-06/</a>

Queensland Department of Health (QDH). Information Sheet. 2.2 Funder/Purchaser/Provider Model. Health System Induction Model No. 2 Funding Arrangements http://www.health.qld.gov.au/ohsa/docs/2-2.pdf (accessed April 2014).

Railkar R, Taddeo FJ, Jansen KU, Esser MT, Sings HL, Saah SJ, Barr E. 2005. Prophylactic quadrivalent human papillomavirus (types 6,11,16 and 18) L1 virus-like particle vaccine in young women: a randomized double-blind placebo-controlled multicentre phase 11 efficacy trial, *Lancet Oncology*. 6: 5: pp271-8.

Rappoport J. 2011. Faking Medical Reality. *Jon Rappoport's Blog: interesting, innovative and investigative reporting.* http://jonrappoport.wordpress.com/2011/05/09/faking-medical-reality/ (accessed Nov 2012).

Reynolds C. 2004. Public Health Law. In Cromar N, Cameron S, Fallowfield H. (eds). *Environmental Health in Australia and New Zealand*. Melbourne: Oxford University Press. pp167-180.

Rogers L. 1984. Pharmacology: why drug prescription is on the increase. In Birke L and Silvertown J. (eds). *More than the Parts: Biology and Politics*. London: Pluto Press. Ch 4.

Rosenthal E. 2008. The Evidence Gap: Drug Makers Push Leads to Cancer Rise. *The New York Times*. August 20. <a href="https://www.nytimes.com/2008/08/20/health/policy/20vaccine.html">www.nytimes.com/2008/08/20/health/policy/20vaccine.html</a> (accessed December 2009).

Rosella LC, Wilson K, Crowcroft NS, Chu A, Upshur R, Willison D, Deeks SL, Schwartz B, Tustin J, Sider D, Goel V. 2013. Pandemic H1N1 in Canada and the use of evidence in developing public health policies - A policy analysis. *Social Science and Medicine*. 83: pp1-9

Rothman SM and Rothman DJ. 2009. Marketing HPV Vaccine: Implications for Adolescent Health and Medical Professionalism. *Journal of the American Medical Association*. 302: 7: pp781–785.

Ryan M, Murphy G, Ryan E, Nilsson L, Shackley F, Gothefors L, Oymar K, Miller E, Storsaeter J, Mills KHG. 1998. Distinct T-cell subtypes induced with whole cell and acellular pertussis vaccines in children. *Immunology*. 93: pp1-10.

Salmon DA, Moulton LH, Halsey NA. 2004. Enhancing Public Confidence in Vaccines Through Independent Oversight of Postlicensure. *American Journal of Public Health*. Washington. 94: 6(June): pp947-950.

Sandberg K and Justice J. 2013. National commitments and global objectives. In Roalkvam S, McNeill D and Blume S (eds). *Protecting the World's Children: Immunisation Policies and Practice*. Oxford. Oxford University Press. pp87-115.

S.A.N.E Vax Inc. Safe, Affordable, Necessary and Effective Vaccines. <a href="www.sanevax.org">www.sanevax.org</a> (accessed April 2011).

Sarojini NB, Sandhya Srinivasan, Madhavi Y, Srinivasan S, Anjali Shenoi. 2010. The HPV Vaccine: Science, Ethics and Regulation. *Economic and Political Weekly*. 45: 48: pp27-34.

Schiffman M and Rodriquez AC. 2008. Heterogeneity in CIN 3 Diagnosis. *Lancet Oncology*. 9: 5: pp404-406

Schiffman M. 2002. In Kircheimer S. Herpes Linked to Cervical Cancer. *WebMD Health News*.

Schwenk TL. 2006. A Look Back at the Recent Smallpox Vaccination Program. *Journal Watch*. www.general-medicine.jwatch.org/cgi/content/full/2006/110/1 (accessed March 2006).

Science and Environmental Health Network (SEHN). The Precautionary Principle. http://sehn.org/precautionary-principle/ (accessed September 2011)

Seedhouse D. 2009. *Ethics: The Heart of Health Care* (Third Edition). Chichester UK: Wiley.

Seife C. 2012. How Drug Company Money is Undermining Science: the pharmaceutical industry funnels money to scientists who are doing research that affects its products – and nobody can stop it. Original Title: Is Drug Research Trustworthy? *Scientific American*. November. <a href="http://www.scientificamerican.com/article.cfm?id=how-drug-company-money-undermining-science&page=9">http://www.scientificamerican.com/article.cfm?id=how-drug-company-money-undermining-science&page=9</a> (accessed Nov 2012).

Sepkowitz KA. 1996. Occupationally Acquired Infections in Health Care Workers Part II. *Annals of Internal Medicine*. 125: pp917-28.

Sepkowitz KA and Eisenberg L. 2005. Occupational Deaths among Healthcare Workers. *Emerging Infectious Diseases*. Center for Diseases Control and Prevention (CDC). 11: 7(July): <a href="http://wwwnc.cdc.gov/eid/article/11/7/pdfs/04-1038.pdf">http://wwwnc.cdc.gov/eid/article/11/7/pdfs/04-1038.pdf</a> (accessed June 2012).

Shaw I and Kauppinen K. 2004. *Constructions of Health and Illness*. Aldershot, Hants: Ashgate

Shoenfeld Y and Agmon-Levin N. 2011. ASIA – Autoimmunity/inflammatory syndrome induced by adjuvants. *Journal of Autoimmunity*. 36: pp4-8.

Slade B A, Leidel L, Vellozzi C, Woo EJ, Hua Wei, Sutherland A, Izurieta H, Ball R, Miller N, Braun M, Markowitz L, Iskander J. 2009. Postlicensure Safety Surveillance for Quadrivalent Human Papillomavirus Recombinant Vaccine. *Journal of the American Medical Association*. 302: 7: pp750–757.

Slovic P. 1998. The Risk Game. *Reliability Engineering and Systems Safety*. 59: 1: pp73-7.

Smith AM. 1999. Analysis and expression of important vaccine antigens of *Bordetella pertussis*. PhD Thesis, University of Wollongong.

Smith P. 2013. Chiros ordered to ditch anti-vax message. *Australian Doctor*. 9<sup>th</sup> August.

Smith R. 2005. Medical Journals are the extension of the Marketing arm of Pharmaceutical Companies. *PloS Med* 2: 5 (May): e138.

http://www.plosmedicine.org/article/info:doi/10.1371/journal.pmed.0020138 (accessed March 2013).

Smith JS, Herrero R, Bosetti C, Munoz N, Bosch FX, Eluf-Neto J, Castellsague X, Meijer CJL, Van den Brule AJC, Franceschi S, Ashley A. For the International Agency for Research on Cancer (IARC) Multicentric Cervical Cancer Study Group. 2002. Herpes Simplex Virus - 2 as a Human Papillomavirus cofactor in the etiology of invasive cervical cancer. *Journal of the National Cancer Institute*. 94: 21: pp1604–1613.

Stacey M. 1988. The Sociology of Health and Healing; A Textbook. London: Unwin Hyman.

Stamatikis E, Weiler R, Ioannidis J. 2013. Undue industry influences that distort healthcare research, strategy, expenditure and practice: a review. *European Journal of Clinical Investigation*. 43: 5(May): pp469-475.

Stanley FJ. 2001. Centenary Article: Child Health Since Federation. In *Yearbook Australia* 2001. Canberra: Australian Bureau of Statistics [ABS Catalogue No. 1301.0]. pp368-400.

Stern AM and Markel H. 2005. The History of Vaccines and Immunization: Familiar Patterns, New Challenges. *Health Affairs: The Policy Journal of the Health Sphere*. 24: 3: pp611-621 http://content.healthaffairs.org/cgi/content/full/24/3/611 (accessed 31/03/06).

Stern PC and Fineberg HV (Eds). 1996. *Understanding Risk: Informing Decisions in a Democratic Society*. Washington DC: National Academy Press.

Stewart G T. 1977. Vaccination against Whooping Cough: Efficacy v Risks. *The Lancet*. 29(Jan): pp234-237.

Stokes B. 2010. Ministerial Review into the Public Health Response into the Adverse Events to the seasonal Influenza Vaccine. Government of Western Australia, Department of Health, August.

Stoneham M, Dodds J, and Buckett K. 2004. Policy in the Government Context. Ch.8 in Cromar N, Cameron S, Fallowfield H. (eds). *Environmental Health in Australia and New Zealand*. Melbourne: Oxford University Press.

Strahan L. 1994. An Oriental Scourge: Australian and the Asian flu epidemic of 1957. *Australian Historical Studies*. 103(Oct): pp182-201.

Sweet M. 2009. The shot of PR that could undermine public health policy. *Crikey*. www.crikey.com.au/2010/09/29/the-shot-of-pr-that-could-undermine-public-health-policy (accessed August 2011).

Szreter S. 1988. The Importance of Social Intervention in Britain's mortality decline c.1850-1914: A reinterpretation of the role of public health. *Social History of Medicine*. 1: pp1-37.

Thomas D, Ray R, Pardthaisong T, Chutivongse S, Koetsawang S, Silpisornkosol S, Virutamasen P, Christopherson W, Melnick J, Meirik O, Farley T, Riotton G. 1996. Prostitution, Condom Use and Invasive Squamous Cell Cervical Cancer in Thailand. *American Journal of Epidemiology*. 143: 8: pp779 – 86.

Tomljenovic L. 2010. Aluminium and Alzheimer's disease: After a Century of Controversy, Is there a Plausible Link? *Alzheimers Disease*. 23: pp1-32.

Tomljenovic L and Shaw CA. 2011. Aluminium vaccine adjuvants: Are they safe? *Current Medicinal Chemistry*. 18: 17: pp2630-2637.

Tomljenovic L and Shaw CA. 2012. Mechanisms of aluminium adjuvant toxicity and autoimmunity in pediatric populations. *Lupus*. 21: 2(Feb): pp221-223.

Tomljenovic L, Wilyman J, Vanamee E, Bark T, Shaw C. 2013. Letter to the Editor. HPV Vaccines and Cancer Prevention: Science versus Activism. *Infectious Agents and Cancer*. 8: 6 (Feb): pp1-3.

Tomljenovic L and Shaw CA. 2013. Human papillomavirus (HPV) vaccine policy and evidence-based medicine: Are they at odds? *Annals of Medicine*. 45 (March): 2: pp182-193.

Tozzi AE, Asturias EJ, Balakrishnan MR, Halsey NA, Law B, Zuber PL. 2013. Assessment of causality of individual adverse events following immunisation (AEFI): a WHO tool for Global use. *Vaccine*. 31: 44(Oct): pp5041-6.

Trifonov V, Khiabanian H, Rabadan R. 2009. Geographic Dependence, Surveillance, and Origins of the 2009 Influenza A (H1N1) Virus. *The New England Journal of Medicine*. 361 (July): pp115-119.

Trottier C, Colombo M, Mann K, Miller WH, Ward BJ. 2009. Retinoids inhibit measles virus through a type 1 IFN-dependent bystander effect. *The Journal of the Federation of American Societies for Experimental Biology*. 23: 9: pp3203-3212.

Uniquest Pty Ltd. The University of Queensland. <a href="https://www.uniquest.com.au/uniquest-corporate-contact">https://www.uniquest.com.au/uniquest-corporate-contact</a> (accessed august 2014).

Uniquest. Technology Investment and Licensing (TIL). Gardasil Human Papillomavirus Vaccine. <a href="https://www.uniquest.com.au/portfolio/gardasil-human-papilloma-virus-vaccine">https://www.uniquest.com.au/portfolio/gardasil-human-papilloma-virus-vaccine</a> (accessed Nov 2011).

United Nations Conference on Environment and Development (UNCED). 1992, Agenda 21: A Blueprint for Survival in the 21<sup>st</sup> Century. United Nations Environment Program, Rio de Janeiro in Stoneham et al. 2004.

Committee on Oversight and Government Reform. 2012. 1 in 88 Children: A Look into the Federal Response to Rising Rates of Autism. US Congressional Hearing November 29.

US National Library of Medicine (NLM). 2002. Greek Medicine; the Hippocratic Oath. National Institutes of Health (NIH). <a href="http://www.nlm.nih.gov/hmd/greek/greek\_oath.html">http://www.nlm.nih.gov/hmd/greek/greek\_oath.html</a> (accessed Nov 2011).

US Environmental Protection Agency (EPA). Toxicity and Exposure Assessment for Children's Health (TEACH). 2007. Organic Mercury: TEACH Chemical Summary. US Government <a href="https://www.epa.gov/teach">www.epa.gov/teach</a> (accessed 2013).

US Vaccine Court. 1986. National Childhood Injury Act in Holland M, Conte L, Krakow R, Colin L. 2011. Unanswered Questions from the Vaccine Injury Compensation Program: A Review of Compensated Cases of Vaccine-Induced Brain Injury. *Pace Environmental Law Review*. 28: 2: pp480-543. http://digitalcommons.pace.edu/pelr (accessed May 2011).

Vaccine Injury Information. The Comparison of Vaccinated versus Unvaccinated <a href="http://vaccineinjury.info/vaccinations-in-general/vaccinatedunvaccinated.html">http://vaccineinjury.info/vaccinations-in-general/vaccinatedunvaccinated.html</a> (accessed May 2013).

Vallet B. 2009. OIE's role in the pandemic influenza H1N1 2009. World Organisation for Animal Welfare (WOAW). April <a href="http://www.oie.int/en/for-the-media/editorials/detail/article/oies-role-in-the-pandemic-influenza-h1n1-2009/">http://www.oie.int/en/for-the-media/editorials/detail/article/oies-role-in-the-pandemic-influenza-h1n1-2009/</a>

Vanlangendonck P. 2014. French Vaccine Debates: What immediate measures are required? June 4 <a href="http://sanevax.org/french-vaccine-debates-immediate-measures-required/">http://sanevax.org/french-vaccine-debates-immediate-measures-required/</a> (accessed June 2014).

van Doorn L-J, Molijn A, Kleter B, Quint W, Colau B. 2006. Highly effective detection of human papillomavirus 16 and 18 DNA by testing algorithm combining broad-spectrum and type-specific PCR. *Journal of Clinical Microbiology*. 44(Sept): pp3292-8.

Vetter KM and Gellor SE. 2007. Moving Forward: human papillomavirus vaccination and the prevention of cervical cancer. *Journal of Women's Health*. 16: 9(Nov): pp1258–1268. doi:10.1089/jwh.2007.0493.

Villa LL and Franco E. 1989. Epidemiologic Correlates of Cervical Neoplasia and Risk of Human Papillomavirus Infection in Asymptomatic Women in Brazil. *Journal of the National Cancer Institute*. 81: 5: pp332-340.

Villa LL, Costa RL, Petta CA, Andrade RP, Ault KA, Giuliano AR, Wheeler CM, Koutsky LA, Malm C, Lehtinen M, Skjeldestad FE, Olsson SE, Steinwall M, Brown DR, Kurman RJ,Ronnett BM, Stoler MH, Ferenczy A, Harper DM, Tamms GM, Yu J, Lupinacci L, Virtanen M, Peltola H, and Heinonen OP. 2000. Day-to-day reactogenicity and the healthy vaccine effect of measles-mumps-rubella vaccination. *Pediatrics*. 106: 5(Nov): E62 PubMed PMID: 11061799.

Waitzkin H. 2003. Report of the WHO Commission on Macroeconomics and Health: a summary and critique. *The Lancet*. 361 (February). pp523-526.

Walboomers J, Jacobs M, Manos M, Bosch X, Kummer J, Shah K, Snijders P, Peto J, Meijer C, Munoz N. 1999. Human papillomavirus is a necessary cause of Invasive Cervical Cancer Worldwide. *Journal of Pathology*. 189: pp12-19.

Walker MJ. 1993. *Dirty Medicine: Science, Big Business and the Assault on Natural Health Care.* London: Slingshot Publications.

Wallace AR. 1898. *The Wonderful Century: its successes and its failures*. New York: Dodd, Mead and Company Publishers.

Warfel JM, Zimmerman LI, Merkel TJ. 2013. Acellular pertussis vaccines protect against disease but fail to prevent infection and transmission in a non-human primate model, Division of Bacterial, Parasitic and Allergenic Products. Center for Biologics Evaluation and Research. US Food and Drug Administration (FDA). Bethesda MD, 20892.

Wendelboe AM, Van Rie A, Salmaso S, Englund JA. 2005. Duration of Immunity Against Pertussis After Natural Infection or Vaccination. *The Pediatric Infectious Disease Journal*. 24: 5: pp558-561.

Williamson S. 2005. HPV vaccine looks promising: Frazer. *Australian Life Scientist*. 9<sup>th</sup> May www.lifescientist.com.au/article/131134/hpv\_vaccine\_looks\_promising\_frazer/ (accessed Nov 2011).

Wilson CB, Westall J, Johnston L, Lewis DB, Dower SK. 1986. Decreased production of interferon-gamma by human neonatal cells: intrinsic and regulatory deficiencies. *The Journal of Clinical Investigation*. 77: 3: pp860-867.

Winkelstein W. 1972. Epidemiological considerations Underlying the Allocation of Health and Disease Care Resources. *International Journal of Epidemiology*. 1: 1: pp69-74.

Witt MA, Katz PH, Witt DJ. 2012. Unexpectedly limited durability of immunity following acellular pertussis vaccination in pre-adolescents in a North American outbreak. *Clinical Infectious Diseases*. 54: 12(June): pp1730-1735.

Woodhead M. 2013. Vax objections rising: GP's not legally obliged to sign conscientious objector forms. *Australian Doctor*. 10<sup>th</sup> May.

World Association for Vaccine Education (WAVE). Vaccine Ingredients and Adverse Events. <a href="http://www.novaccine.com/vaccine-ingredients/">http://www.novaccine.com/vaccine-ingredients/</a> (accessed March 2013).

World Medical Association. 1948. Declaration of Geneva <a href="http://www.wma.net/en/30publications/10policies/g1/">http://www.wma.net/en/30publications/10policies/g1/</a> (accessed June 2015)

World Health Organisation (WHO). 2007. Global Influenza Program (GIP). Report of the WHO Consultation on Surveillance for Pandemic Influenza. Geneva. Switzerland. 10-12 Dec.

World Health Organisation (WHO). 2008. Human Papillomavirus (HPV) Vaccine Background Paper. Biotext Pty Ltd, Canberra, September

World Health Organisation (WHO). 2008. WHO: International Health Regulations (IHR) 2005. Geneva.

World Health Organisation (WHO). 2008. National Immunisation Technical Advisory Group (ITAG): Guidance for their establishment and functioning. 23 September. <a href="http://www.who.int/immunization/sage/National\_TAG\_guidelines\_23September\_2008.pdf">http://www.who.int/immunization/sage/National\_TAG\_guidelines\_23September\_2008.pdf</a> (accessed August 2015).

World Health Organisation (WHO). 2008. Preparing for the Introduction of HPV Vaccine in the WHO European Region; Strategy Paper, Vaccine-Preventable Diseases and Immunisation Program. WHO Regional Office for Europe. Denmark.

The World Health Organization (WHO). 2009. Global Alert and Response: What is the pandemic (H1N1) 2009 virus? (updated February 2010) (accessed June 2009 and February 2010).

www.who.int/csr/disease/swineflu/frequently\_asked\_questions/about\_disease/en/index.html

World Health Organisation (WHO). Global Influenza Surveillance Network (GISN). <a href="https://www.who.int/">www.who.int/</a> (accessed November 2010).

World Health Organisation (WHO). 2009(i). WHO use of advisory bodies in responding to the influenza pandemic, Pandemic (H1N1) 2009 briefing note 19.

World Health Organisation (WHO). 2009(ii). Considerations for Assessing the severity of an influenza pandemic. *Weekly Epidemiological Record*. 22 (May): 84: pp197 – 212.

World Health Organisation (WHO). 2009(iii). List of Members of, and Advisor to, the International Health Regulations (2005) Emergency Committee concerning Influenza Pandemic (H1N1) 2009. www.who.int/ (accessed August 2010).

World Health Organisation (WHO). 2009(iv). Human infection with new influenza A(H1N1) virus: clinical observations from mexico and other affected countries. *Weekly Epidemiological Record*. 21: 84 (May): pp185-196.

World Health Organisation (WHO). 2010. WHO: progress in public health during the previous decade and major challenges ahead. Dr. Margaret Chan, DG, Report to the Executive Board at its 126<sup>th</sup> session. Geneva. Switzerland.

## World Health Organisation (WHO):

- a) Chronic Diseases (CD). <a href="http://www.who.int/topics/chronic\_diseases/en/">http://www.who.int/topics/chronic\_diseases/en/</a> (accessed March 2013).
- b) Commission on Social Determinants of Health (CSDH). 2005. Action on the Social Determinants of Health: Learning From Previous Experiences.
- c) Member Countries (updated 15<sup>th</sup> October 2012). http://www.who.int/countries/en/ (accessed September 2013).
- d) Health Promotion. The Ottawa Charter for Health Promotion 1986. http://www.who.int/healthpromotion/conferences/previous/ottawa/en/index1.html (accessed November 2012).
- e) Immunisation Service Delivery (ISD). Expanded Program on Immunisation (EPI). (updated 15<sup>th</sup> October 2013). http://www.who.int/immunization\_delivery/en/\_\_(accessed September 2013).
- f) Preamble to the Constitution of the World Health Organization as adopted by the International Health Conference. 1946. Signed on 22 July 1946 by the representatives of 61 States (Official Records of the World Health Organization, no. 2, p. 100) and entered into force on 7 April 1948. New York. http://www.who.int/about/definition/en/print.html (accessed November 2012).
- g) Media Centre. Immunisation Coverage (IC). Fact Sheet No. 378 (updated November 2013) <a href="http://www.who.int/mediacentre/factsheets/fs378/en/index.html">http://www.who.int/mediacentre/factsheets/fs378/en/index.html</a> (accessed November 2013)
- h) Swine Flu Frequently Asked Questions. (updated 11<sup>th</sup> June 2009) <a href="https://www.who.int/csr/disease/swineflu/frequently\_asked\_questions/about\_disease/en/index.html">www.who.int/csr/disease/swineflu/frequently\_asked\_questions/about\_disease/en/index.html</a> (accessed September 2009).

Zeigler JB, Burgess M, Gilbert G, McIntyre P. 1991. The Australian College of Pediatrics Policy Statement. Report of the immunization subcommittee on pertussis immunization, *Journal of Pediatrics and Child Health.* 27: 1: pp16-20.

Ziman J. 2000. Real Science. Cambridge. UK: Cambridge University Press.

zur Hausen H. 2008. Nobel Media AB. *Nobelprize.org*. <a href="http://www.nobelprize.org/nobel\_prizes/medicine/laureates/2008/hausen-bio.html">http://www.nobelprize.org/nobel\_prizes/medicine/laureates/2008/hausen-bio.html</a> (accessed July 2014).